\\ 195 \\

Crossing the Rio Grande: Migrations, Business Cycle and the Welfare State

by

Fabio Canova* Morten O. Ravn **

August 1997

- * Università degli Studi di Modena Dipartimento di Economia Politica Viale Berengario, 51 41100 Modena (Italia) e – mail: <u>canova@unimo.it</u>
- ** University of Southampton Universitat Pompeu Fabra and CEPR

ka/ **9**07 1. N

27

Abstract

In this paper we study the macroeconomic effects of an inflow of low-skilled workers into an economy where there is capital accumulation and two types of agents. We find that there are substantial dynamic effects following unexpected migrations with adjustments that resemble those triggered by a sudden disruption of the capital stock. We look at the interrelations between these dynamic effects and three different fiscal systems for the redistribution of income and find that these schemes change the dynamics and lead to prolonged periods of adjustments. The aggregate welfare implications are sensitive to the welfare system: while there are welfare gains without redistribution, these gains are turned into costs when the state engages in redistribution.

^{*}Canova gratefully acknowledges the financial support of DCIGYT and CNR grants. We would like to thank participants of the HCM Conference on "Growth and Business Cycles in the EU", Hydra, April 1996, of the workshop on "The Welfare State: Threats, Problems and Some Solutions", Ebeltoft, Denmark, June 1996, and of seminars at Birkbeck, LSE, UPF, IEA, European Forum of the EUI and especially Tryphon Kollintzas, Finn Kydland, Lisa Lynch and Harald Uhlig for comments. The hospitality of the European Forum at the EUI is gratefully acknowledged. The first author is also associated with Universitat Pompeu Fabra and University of Southampton.

8

1 Introduction

Migration flows from underdeveloped areas to OECD countries have been substantial in the past and even over the last decade they have continued to play an important role in the demographics of many industrialized countries. For example, in the 16 EU-member countries, net immigration has been about 0.5% of the native population in every of the last ten years and this factor has been by far the most important source of aggregate population growth. There have also been recent episodes of large and temporarily concentrated migration flows, e.g. Albanians to Italy and Greece or Eastern Europeans to Central Europe at the beginning of the 90s, and one could think of German unification as a mass migration from an underdeveloped country towards a large and developed one.

Both policy circles and the popular press seem to be particularly concerned with migrations and there have been international actions to prevent mass migrations toward industrialized nations: political pressure has led to regulate the flow and qualifications of migrants (e.g. in Australia and Canada), and to restrict the type of labor activities migrants can undertake and the benefit they enjoy. The latest example of this type is the US Illegal Immigration Reform and Immigration Responsibility Act passed in 1996 which exclude legal migrants from certain welfare benefits (Supplementary Security Income, Food stamps and, in some cases, Medicaid) enjoyed by natives or naturalized citizens, disregarding the fact that these migrants might have payed social security taxes in the past. It has been estimated (see Super and Daskal (1997)) that such a change could affect 450.000-700.000 legal migrants in the US, little less than 0.5% of the population, with budget savings for the welfare state of about 1.5 billion dollars.

In this paper we attempt to shed some light on why public perception toward migrations in OECD countries has changed in the last 15-20 years by answering the following questions: (i) What are the dynamic effects of an inflow of agents in the economy? (ii) Are these effects dependent on the skill composition of the migrants relative to the native population? (iii) Do migration flows affect sectors of native population differently? Which group is more affected? (iv) Do macroeconomic effects depend on the presence of a welfare state? (v) How do restrictions on migrants' access to the labor market and the welfare state alter the macroeconomic effects of labor flows across countries?

We consider a model of a single economy and neglect both the economic consequences on the sending country and the international repercussions. This seems a sensible modelling strategy once it is taken into account that the focus of our analysis is a large and developed receiving country, which one may think of as Europe or the US, and that migrations take place primarily because of political reasons. Furthermore, since (at least until very recently) most migrants are low-skilled relative to the average native our model has heterogeneous agents differing in their productivity: the native population is initially made up of two types of agents, and the highly-skilled type is more productive than the low-skilled type. In modelling this feature we highlight the fact that the degree of substitutability between the two types of labor is an important factor determining both the

1 INTRODUCTION

aggregate repercussions and the distributional effects on natives of migration flows.

Within this set-up we model migrations as a temporary increase in the growth rate of the unskilled portion of the population and examine the consequences of such an inflow under different redistribution schemes. Initially, we assume that there is no redistribution and study whether migrations with different features produce different dynamics responses of aggregate and sectorial variables. This may be thought of as a model for the US or for Europe before the establishment of the various components of the welfare state. Afterwards we consider a constant income tax rate system with tax receipts redistributed to low-skilled workers. The third case we look at is related to the system currently in place, e.g., in Canada or the Scandinavian countries where redistributions are guided by egalitarian considerations. Finally, we look at a case where the tax system is used to insure low-skilled consumption. This system bears resemblance to the insurance scheme in place after German unification (see Schrettl (1992)) and it is foreseen in analyses of the future common fiscal unit of the European Union.

We find that the presence of a welfare state is important for evaluating how migrations affect the economy and that it may be one of the factors explaining the change in the public perception towards migrations. Without redistributive policies, an inflow of low-skilled labor triggers an adjustment process that resembles that would take place if the capital stock was suddenly disrupted: since the population growth rate temporarily increases, there is initially less capital per-capita and agents work harder to rebuild the capital stock. This process leads initially to a downward pressure on wages and an increase in interest rates. Indeed, if all workers were identical and no redistribution was in place, this would be the only adjustment and there would be no long-run effects. However, with heterogeneous agents, there is a second effect since the composition of the population changes permanently. This compositional effect causes both more substantial short-run dynamics and permanent long-run effects. In particular, if migrants have lower productivities than the average native worker and the two types of labor are sufficiently substitutable in production, steady-state output, capital, hours in efficiency units and high-skilled hours will be permanently lower in the new steady-state.

It is worth noting that our set-up, which allows for capital accumulation, leads to dynamics which are different than those typically encountered in models with a fixed capital stock (see e.g. Benhabib (1996)). In such models an inflow of labor typically leads to a once-and-for-all change in the capital-labor ratio: if migrants carry less capital than the average native the capital-labor ratio decreases, thereby making capital owners better off and workers worse off and vice versa if the migrants carry more capital than the average native. These modifications roughly correspond to the short-run effects we find in our set-up. However, because of capital accumulation, long-run effects will be different in our model. In the long-run, the capital-labor ratio is determined by preferences, technology and taxes via a modified golden rule. Thus, if taxes are unaltered and the two types of workers perfect substitutes, a migration flow does not affect wages and interest rates in the long run and simply leads to an increase in the wealth of capital owners who end up holding more capital per-capita. Obviously, if taxes change or the two types of workers are imperfect substitutes, these long run effects will be altered.

An inflow of low-skilled labor produces important redistributive effects from wage to capital income along the adjustment path since the wage rate is depressed while the return to capital increases. In addition, if the two types of workers are imperfect substitute in production, such an inflow produces a further redistribution from low-skilled to highskilled agents. Given these effects we find that, if we take skill differences between natives and migrants as exogenous, transient migrants may be preferred to permanent ones if the objective is to minimize long run changes and that high-skilled migrations may be beneficial for the economy both over the business cycle and in the long-run.

All these effects, however, depend on the fiscal system. For example, high-skilled/capitalists may be heavily penalized by the welfare state under certain policies because they have to support a larger number of low-income agents. Essentially, redistribution schemes can shift the costs of migrations from low-skilled to high-skilled/capitalists providing additional depressive effects on capital accumulation both over the business cycle and in the long run. We show that these effects may be partially avoided if the admittance of migrants to the welfare state provisions is limited but it is hard to believe that this policy will be politically sustainable in the medium-long run.

As a corollary to our analysis, we consider the effects of an inflow of illegal migrants and of restrictions on the labor market activities of migrants. In our set-up the distinguishing feature of illegal migrants is that these agents do not receive transfers and do not pay income taxes. When the state is egalitarian we find, maybe surprisingly, that both types of agents would encourage illegal immigration! The reason for this is that low-skilled agents are net-receivers of transfers from the welfare state. Thus, excluding these migrants from the welfare state means that taxes need to be increased less after an inflow of low-skilled agents, which is a policy favored by the high-skilled agents, and that the tax revenues are redistributed to the native low-skilled agents only. Another surprising result is that, with a welfare state, low-skilled natives favor the admittance of migrants to the labor market while skilled agents will oppose it. The reason is that when the migrants are allowed to work the welfare state will be more prone to redistribute, which harms high-skilled agents and improve the welfare of low-skilled agents.

Finally, we attempt to quantify the social welfare costs of migrations. We compute the percentage change in consumption, evaluated from one steady-state to another, which keeps utility unchanged before and after the flow of low-skilled agents takes place. Two measures are investigated: the first measures the welfare changes if capital owners were to bear the entire costs or benefits of migrations, while the second measure weights the utility of the two types of native agents. When the economy does not engage in redistribution, we find that there are aggregate welfare gains from migrations. For example, the economy would be willing to give up 0.25-0.6% of annual consumption to encourage low-skilled migrations which change the composition of population by 1%. This result is well-known in models with fixed capital stocks, reappears here and it is due to the fact that capital owners are better off. With a welfare system in place, however, these gains are turned into substantial costs (up to 0.5% of annual consumption). The quantitative features of the results depend on the size of migration flows, the skill composition of native population and the elasticity of substitution of the two types of labor in production.

The remainder of the paper is organized as follows: section 2 discusses relevant facts about recent migration flows to the OECD countries; section 3 outlines our model; section 4 analyzes the quantitative effects of migrations; section 5 repeats the analysis under various welfare systems; section 6 computes the social welfare costs of migrations and section 7 concludes.

2 Some Facts about Recent Migrations in OECD Countries

We begin by providing some facts about migration flows in recent years. Far from being complete, our description outlines primarily the facts which are useful to build a sufficiently articulated dynamic model where migrations generate interesting effects. For a thorough description of recent trends we invite the reader to consult "Trends in International Migration", published annually by the OECD since 1992, where we have taken most of the information.

Migration flows toward the US and Europe levelled off in the 90's after the large increase in the trend in the last part the 1980's due to the breakdown of the communist bloc and to worsening economic conditions in the Central and South America and the Southern part of the Mediterranean basin. Over the last decade Germany has been the European country with the largest net immigration flow. In 1992 and 1993 the net flow of legal migrants (excluding foreigners of German background) was about 575,000 and 280,000 which correspond to 0.9% and 0.4% of the existing population. For France and Switzerland there were net immigrations of approximately 100,000 persons each year while for the U.K., Belgium and the Netherlands the net number of legal immigrants was about 50,000 Bosnians each year. Sweden occupies a special place having allowed an inflow of 50,000 Bosnians each year since the beginning of the war in ex-Yugoslavia. In percentage, net migration flows have accounted on average for about 0.5% of the EU population over the last 10 years. In 1993, new inflows as percentage of population were 0.5% in Switzerland, 0.6% in the Netherlands, 0.3% in Norway and Denmark and 0.2% in the UK and France.

On average, 30% of the migrants are workers, most of which are males 14-49 years of age. The rest is composed of accompanying family members, migrants for purpose of family reunification and refugees or asylum seekers. Switzerland, Germany and France are the countries where the percentage of workers among migrants is largest. In particular, in Switzerland in 1993, 43% of the migrants were workers. In the US and Canada, the percentage of workers is low (about 10%) and family reunification accounts for the largest percentage of net inflows.

4

2 SOME FACTS ABOUT RECENT MIGRATIONS IN OECD COUNTRIES

Migrants to Europe come from a variety of continents and countries and over the last decade there has been an increase in the range of nationalities. North African immigration is strong in Spain, France, Italy and Germany; migrants from the ex-Yugoslavia primarily go to Germany, Sweden, Italy and Switzerland; migrants from the old eastern block settle primarily in Germany, Finland, France and Italy; Turks; Middle Easterns prefer Germany, Belgium, France and the Netherlands and finally, East Asians settle primarily in Southern Europe, and in particular, Italy, where most of the east Asian migrants are females. Migrants to the US and Canada are primarily from Asia and Central America.

A large percentage of the migrants toward Central Europe (Germany, Switzerland and Austria) are temporary or seasonal workers who remain in the country for six months or less. For example, in 1993, 200,000 migrants to Germany and 72,000 to Switzerland were seasonal. Also, while at the beginning of the 90's migration flows from Eastern to Central Europe had permanent characteristics, since 1993 there has been a shift from permanent to temporary, seasonal or contract-based migrations. A similar pattern seems to occur in the US, Canada and Australia. Most of the migrants are low-skilled but the percentage of migrants with little qualification has being declining over the last decade. Movements of highly qualified labor are still very small but increase at a sufficiently rapid rate.

For the countries in the EU, net migration is by far the predominant component of the growth of population over the last 10 years. For example, in 1992-93, 68% of the growth in the European population was due to migrations (59% in the Nordic countries). Over the 1987-93 period local growth rates of population were around 0.2% while the migrant population growth rate was approximately 0.5%. Italy is the most extreme case in this respect, followed by Germany: the growth rate of the local population was 0.05% and the one due to migration was 0.3%. For the US and Canada the share of population growth due to migrations is smaller but still substantial (on average 35%).

Migration flows to EU (and OECD) countries over the 80's and 90's appears to be unrelated to business cycle conditions and, instead, linked to political and economic changes. The share of foreigners in the labor force has dramatically increased in the 1990-93 period as a result of migrations, changes in the participation rate of nationals and demographics. In EU countries migrants have unemployment rates which are similar to those of nationals while in the US and Canada foreign-born workers have higher and more variable unemployment rates than nationals and are more vulnerable to unemployment spells. The increase in the number of arrivals in the 90's contributes to the swelling of unemployed foreigners, particularly for low-skilled workers. Over the period 1983-92 foreign employment has increased faster than native employment in half of the OECD countries and the increase was spread to all sectors (including tertiary and services).

In Europe foreign (male) labor is concentrated in sectors which are very cyclical and are either declining or undergoing restructuring. In Germany, for example, foreign workers are most represented in manufacturing, where they account for 15% of total employment; in France they account for about 18% of employment in construction; in Belgium and the Netherlands they account for 15% and 8% of employment in extraction. In the US

foreigners account for about 25% of the employed in precision engineering and other hightech industries. Also, laid-off foreign workers are typically 50+ years old and have held manual and unskilled occupations for a number of years, a characteristic which gives them little chance of being re-employed in the upturn of the business cycle.

3 The Basic Model

The choice of a model, which we use to analyze how an inflow of labor affects an economy, is guided by a number of considerations. First, we want to concentrate on the consequences for the receiving country and we ignore both international repercussions and the effects on the sending country. Hence, we will be concerned with a single and large economic unit such as Europe or the US. Second, we wish to examine the dynamic adjustment of macro variables following labor migrations of different types and with different persistence characteristics. Thus, we will employ a standard neo-classical growth model with capital accumulation. Third, we want to consider a native heterogeneous labor force since we are particularly interested in modelling an inflow of workers whose skill composition differs from the one of natives. Hence, we will work with an economy with two types of agents that differ in their productivity levels (as in Kydland (1984, 1995) and Rios-Rull (1993)) and we follow Kydland in modelling productivity differences.

We assume that low-skilled agents are unable to save for future occurrences. The fact that low-income households are restricted from accessing financial markets is well documented in the literature. For example, Campbell and Mankiw (1989) and Mankiw and Zeldes (1991) have estimated that approximately 50% of US households are liquidity constrained. A large proportion of migrants who move to Europe or the States owns little or no capital and are de-facto restricted from engaging in borrowing activities so they may be thought of as liquidity constrained. However in this paper, we go a step further and assume that low-skilled agents are unable to intertemporally smooth consumption. This assumption is partly made for computational ease but we believe that it is reasonable for at least two reasons. First, in the US about 25% of the households hold either negative or no wealth and the head of the household is typically low-skilled. In European countries this number varies from 5 to 20% but it is still true that, typically, the head of households with no wealth has low skills. Secondly, first generation low-skilled migrants transfer a large portion of their income back to their home country at regular periods and tend to live close to the subsistence level. Moreover, because they are less wealthy than the average native, they are typically unable to purchase insurance contracts or assets which would allow them to smooth their consumption stream.

It is instructive to present the economy first and then discuss the modelling of migration. We let the index i = u, s, denote the two group of agents (s denotes high-skilled and u denotes low-skilled). The number (measure) of agents of type i at time t is given by N_t^i and we define γ_t as the time t population share of high-skilled agents.

We assume that there is a large number of identical competitive firms renting factors

3 THE BASIC MODEL

of production from the households. Production takes place using labor in efficiency units (H^e) and capital (K) and we assume that the production function is Cobb-Douglas with constant returns to scale to these two factors. Hours in efficiency units is modelled as a CES-aggregate of the two types of labor with an elasticity of substitution of $1/\rho$. We also assume that high-skilled hours are more productive than low-skilled hours and we let $\omega \geq 1$ denote productivity differences. When these two factors are perfect substitutes $H^e = H^u + \omega H^s$, where H^i denotes total number of hours of workers of type *i*.

The maximization problem of the representative firm is given by:

$$\max_{\{H_t^s, H_t^u, K_t\}} z_t \left[\omega(H_t^s)^{1-\rho} + (H_t^u)^{1-\rho} \right]^{\frac{\alpha}{1-\rho}} K_t^{1-\alpha} - w_t^s H_t^s - w_t^u H_t^u - r_t K_t$$
(1)

where $H_t^i \equiv N_t^i \cdot h_t^i$, h_t^i is the number of hours of a worker of skill i^1 , and z_t is a productivity disturbance. Converting (1) in per-capita terms using our definition of γ_t , we obtain:

$$\max_{\{h_t^s, h_t^u, k_t\}} z_t \left[\omega(\gamma_t h_t^s)^{1-\rho} + ((1-\gamma_t)h_t^u)^{1-\rho} \right]^{\frac{\alpha}{1-\rho}} (\frac{k_t}{g_t^p})^{1-\alpha} - \gamma_t w_t^s h_t^s - (1-\gamma_t) w_t^u h_t^u - r_t (\frac{k_t}{g_t^p})$$
(2)

where all lower case letters denote per-capita variables and g_t^p the growth rate of the population. It follows immediately from the assumptions of competition that all factors are paid their marginal products so that:

$$w_t^s = \omega w_t^u \left(\frac{\gamma_t h_t^s}{\left(1 - \gamma_t\right) h_t^u}\right)^{-\rho}$$

The two groups of agents face different optimization problems because low-skilled workers own no capital or other assets and that they are restricted from purchasing insurance contracts. These agents solve the following problem:

$$\max_{\{c_t^u, l_t^u\}} E_0 \sum_{t=0}^{\infty} \beta^t U\left(c_t^u, l_t^u\right) \tag{3}$$

subject to a sequence of budget constraints:

$$c_t^u \le w_t^u h_t^u (1 - \tau_t^u) \tag{4}$$

and time constraints:

$$l_t^u + h_t^u \le 1 \tag{5}$$

¹This implicitly assumes that all agents of the same type work the same number of hours. We write it like this, however, only to save on notation.

3 THE BASIC MODEL

where c^u denotes consumption of goods, l^u is leisure, h^u is hours worked, τ^u is the net marginal income tax rate (including transfers), β is the subjective discount factor and where we have normalized the time endowment to one each period. Assuming that the utility function is of the CRRA type, $U(c_t^u, l_t^u) = \frac{1}{1-\sigma^u} \left(\left[(c_t^u)^{\theta^u} (l_t^u)^{1-\theta^u} \right]^{1-\sigma^u} - 1 \right)$, the solution to the problem is:

$$h_t^u = \theta^u, \quad y_t^u = \theta^u \cdot w_t^u, \quad c_t^u = (1 - \tau_t^u) \cdot y_t^u$$

High-skilled workers are more productive than low skilled workers and have access to asset and insurance markets which allow them to smooth consumption over time. Their problem is:

$$\max_{\{c_t^s, h_t^s, k_t\}} E_0 \sum_{t=0}^{\infty} \beta^t U(c_t^s, l_t^s)$$
(6)

subject to a sequence of budget constraints:

$$c_t^s + x_t / \gamma_t \le (w_t^s h_t^s + r_t k_t / \gamma_{t-1})(1 - \tau_t^s)$$
(7)

time constraints:

$$l_t^s + h_t^s \le 1 \tag{8}$$

and the capital accumulation equation:

$$k_{t+1} = (1 - \delta)\frac{k_t}{g_t^p} + x_t$$
(9)

where x_t denotes investments, τ^s the tax rate on skilled agents and δ is the depreciation rate ². If we assume that the utility function is of CRRA type, the intertemporal and intratemporal optimality conditions are :

$$\frac{1 - \theta^s}{\theta^s} \frac{c_t^s}{1 - h_t^s} = w_t^s (1 - \tau_t^s)$$
$$\lambda_t = \beta E_t \lambda_{t+1} \left((1 - \delta) / g_{t+1}^s + (1 - \tau_{t+1}^s) r_{t+1} \right)$$

where λ is the multiplier on the budget constraint (7), g_{t+1}^s is the growth rate of the high-skilled population, and θ^s is the share parameter that enters as the power of consumption in the utility function. The first of these conditions equates the marginal rate of substitution between consumption and leisure to the relative price, while the second condition gives a standard Euler equation.

²Equation (9) can be derived using $k_t^s N_{t-1}^s = K_t$, $k_t = \frac{K_t}{N_{t-1}} = k_t^s \gamma_{t-1}$ and $\frac{\gamma_{t+1}}{\gamma_t} = \frac{g_{t+1}^s}{g_{t+1}^s}$

4 THE DYNAMIC EFFECTS OF LABOR MIGRATION

We assume that there is a government whose aim is to redistribute income across classes of agents via taxes and transfers and it is forced to do this by balancing its budget on a period-by-period basis. We assume that $\tau_t^u = \tau_t^s - \mu_t$, so that μ_t can be thought of as a tax rebate on low-skilled workers, that τ_t^s is either a parameter or is endogenously chosen to target a certain redistributive policy, while μ_t is endogenously determined to satisfy the government budget constraint:

$$\tau_t^s \gamma_t y_t^s = (\mu_t - \tau_t^s)(1 - \gamma_t) y_t^u \tag{10}$$

where y_t^s and y_t^u are the taxable incomes of the two types of agents.

In the aggregate the following income composition and resource constraints must hold:

$$y_t = \gamma_t y_t^s + (1 - \gamma_t) y_t^u \tag{11}$$

$$y_t = \gamma_t c_t^s + (1 - \gamma_t) c_t^u + x_t \tag{12}$$

Finally, we assume that z_t , is first-order autoregressive process,

$$z_t = (1 - \psi_z)\overline{z} + \psi_z z_{t-1} + \varepsilon_t \tag{13}$$

where $\psi \in (0,1)$ and $\varepsilon_t \sim N(0, \sigma_{\varepsilon}^2)$.

4 The Dynamic Effects of Labor Migration

We begin by considering the case of migrations of low-skilled agents with no government and, therefore, set all tax parameters to zero. This step is useful in order to understand how migration disturbances affect the business cycle properties of an economy in isolation from issues of redistribution and welfare. Since most of the migrations in the 1950's and the 1960's occurred in a situation where the welfare state was small or non-existent, our comparative analysis may highlight why public perception toward migrations has changed dramatically over the last 10-15 years and why countries engaged in active policies designed to curb or select the type of migrants allowed in the country.

We model low-skilled migrations as an exogenous temporary increase in the growth rate of low-skilled workers, N_t^u . While it is easy to endogenize migrations as a function of business cycle conditions, we neglect this possibility since it appears to be of minor importance, at least for OECD countries. The temporary increase in number of low-skilled workers has two demographic effects. First, the aggregate population growth rate rises directly. This effect alone produces only short run dynamics since the long-run capitaloutput ratio is determined by the parameters of technology and preferences and none of the other steady-state conditions is affected. Second, there is a permanent effect on the composition of the population which alters long-run conditions.

4 THE DYNAMIC EFFECTS OF LABOR MIGRATION

These two effects can be worked out as follows. Assume (without loss of generality) that the growth rate of the high-skilled population, g_t^s , is constant over time and equal to g^s and that the growth rate of the low-skilled population is $g_t^u = (1 + e_t) g^s$, where e_t is possibly an autocorrelated process. Define the following recursive variable:

$$\Pi_t \equiv \prod_{i=1}^t (1+e_i) = (1+e_t) \Pi_{t-1}, \qquad \Pi_0 = 1.$$
(14)

It then follows that:

$$g_{t+1}^{p} = \frac{(1-\gamma_{0}) \prod_{t+1} + \gamma_{0}}{(1-\gamma_{0}) \prod_{t} + \gamma_{0}} g^{s}$$
(15)

$$\gamma_{t+1} = \frac{\gamma_0}{(1 - \gamma_0)\Pi_t + \gamma_0}$$
(16)

Hence, the growth rate of population is stationary as long as e_t is stationary, while there is a permanent effect on the composition of the population regardless of the assumed process for e_t . This implies that migration flows create short-term and long-run changes in the economy with the latters due to the fact that (low-skilled) migrants can not invest to improve their productivities since skill differences are exogenous in the model.

Before examining the short-run dynamics it is useful to consider long-run effects. As mentioned, a temporary inflow of low-skilled workers permanently decreases the share of high-skilled workers in the population which in turn alters hours in efficiency units and may affect the capital-output ratio (or, alternatively, on the capital-labor ratio). In appendix A we describe in details how these two variables change as a function of various parameters. Here we summarize the main features of the results. The steady-states levelof output, efficiency hours and the capital-output ratio are given by:

$$y = z^{1/\alpha} \left(\frac{k}{y}\right)^{(1-\alpha)/\alpha} (h^e)^{1/(1-\rho)}$$
$$h^e = \omega \left(\gamma h^s\right)^{1-\rho} + \left((1-\gamma)\theta\right)^{1-\rho}$$
$$\frac{k}{y} = \frac{\beta g^p \left(1-\alpha\right) \left(1-\tau^s\right)}{1-\beta \left(1-\delta\right)}$$

where variables without time subscripts denote steady-state levels.

Note that the capital-output ratio is determined by preferences, technology and by the tax-rate levied on high-skilled agents. Thus, the capital-output ratio is affected only if the tax rate on high-skilled agents change with an inflow of low-skilled workers. If taxes are held constant, changes in aggregate activity are exclusively determined by the changes in

efficiency hours. When the two types of labor are perfect substitutes, hours in efficiency units are *increasing* in γ as long as $\omega > 1$. Thus, an inflow of low-skilled agents (a decrease in γ) lowers per-capita output, investment and consumption in the long run. This effect on efficiency hours is solely determined by the effect on high-skilled hours which decrease because of a positive wealth effect. Native low-skilled agents are unaffected by migrations in the long-run because the wage rate is constant

When the two types of labor are imperfect substitutes these results crucially depend on the value of ρ , γ and ω . In the Appendix we show the following results. First, independently of ω and the initial share of skilled workers, skilled hours are increasing in γ for $\rho < 1$ and decreasing in γ for $\rho > 1$. Thus, whether or not skilled hours increase or decrease depends solely on whether the elasticity of substitution exceeds 1 or not. Second, for a sufficiently high initial share of skilled workers, hours in efficiency units is decreasing in γ for all $\rho \ge 0$. If, on the other hand, the initial share of skilled workers is moderately low, hours in inefficiency units may be decreasing in γ and this is more likely the lower is the elasticity of substitution between the two types of labor. For such cases, an inflow of low-skilled workers can increase activity because skilled workers increase their labor supply.

In the standard case (i.e. for sufficiently high elasticities of substitution and/or high initial shares of skilled workers) the steady-state capital stock and output per capita decrease following an immigration of low-skilled workers; high-skilled workers work fewer hours, invest less but have higher income and consumption (because capital income is higher), while low-skilled workers earn lower income and consume less in the long-run because their relative wage declines.

If taxes are varied in response to migrations, the steady-state capital-output ratio is altered since marginal taxes distort the accumulation of capital. Since an inflow of lowskilled workers is likely to be associated with an increase in the size of the welfare state and thus in τ^s , this creates an additional source of negative long-run effects in addition to those coming through changes in hours in efficiency units.

4.1 Calibration

We calibrate the model to match annual data and try to use standard parameter values whenever it is possible. The depreciation rate is set equal to 10% and the interest rate to 4%. The parameter α (the labor share of income) is set to 64%. We also assume that the population growth is stationary in the steady-state. For moderate values of the income tax parameter these values imply a capital-output ratio close to 2.5. We set the intertemporal rate of substitution to 0.5, and we also assume that θ is equal to 0.3. This implies that low-skilled workers use 30% of their non-sleeping time on market activities. In general, high-skilled workers will not use 30% of their time on market activities because they are more productive, own the capital stock and face a different tax rate.

The productivity differential ω is a parameter for which we have less information.

Kydland (1984) and Rios-Rull (1993) suggest a value of 2. Here we use $\omega = 1.3$ in order not to exaggerate the difference between the two types of workers since low-skilled agents hold no capital. We assume that the share of low-skilled agents is 25%. This is consistent with our estimates of low-skilled agents holding negative or zero wealth from the PSID and informal calculations performed on some European countries.

Finally, we need to calibrate the elasticity of substitution between high and low-skilled workers, $1/\rho$. There are many estimates of this parameter in the literature depending on how workers are divided. The closer classification is probably the one of Bean and Pissarides (1991) who divide workers into manual and non-manual in UK manufacturing industries. Their point estimates of the elasticity of substitution vary between 0.2 and 2 with large standard errors, so that one can not reject the hypothesis that it is equal either to zero or to one. Because of this large range we examine two cases: one where the two types of workers are perfect substitute ($\rho = 0$) and another where there is a moderate degree of substitutability between the two types of labor ($\rho = 0.25$).

As a preliminary exercise we have studied how the economy adjusts following an unexpected 1% increase in productivity. Since the aggregate dynamics are very similar to those obtained with a representative agent business cycle model and because there are no surprises at disaggregated level either, we confine the discussion to Appendix B.

In what follows, we examine the dynamic adjustment following a temporary but persistent migration shock which changes the composition of the population in the steady-state from 75% to 73%. We model the e_t as an autoregressive process $e_t = \psi_e e_{t-1} + v_t$ where v_t is i.i.d. $N(0, \sigma_v^2)$. We set ψ_e , which measures the persistence of the migration process, to 0.6 so that current inflows signal future inflows of the same sign. Figure 1 presents the dynamics with perfectly substitutable labor and Figure 2 the dynamics with imperfectly substitutable labor. Baseline parameters values are summarized in the table 1.

Table 1: Baseline Parameters

α	ρ	δ	θ	σ	β	τ^s	μ	g^p	γ	ψ_e	ω
0.64	0	0.1	0.3	2	0.96	0	0	1	0.75	0.6	1.3

4.2 Dynamics with perfect substitutability

The response of aggregate variables is similar, in some respects, to the adjustments brought about by a sudden disruption of the capital stock. On impact an inflow of labor decreases the capital-labor ratio (measured in efficiency units) and this causes the interest rate to increase and the wage rate to decrease. Since the return to capital is high, investments and the aggregate capital stock increase, aggregate consumption decreases and high-skilled agents work more and lower their consumption in order to rebuild the capital stock. Output (per-capita) decreases on impact because the rise in skilled hours is not sufficient to counteract the decrease in the capital-labor ratio.

4 THE DYNAMIC EFFECTS OF LABOR MIGRATION

Over the adjustment path, the wage per efficiency unit decreases (because of the initial drop in the capital-labor ratio) and then returns to its original steady-state value and the interest rate increases and then returns to the initial level in the long run from above. The process is relatively slow with significant adjustments still taking place ten years after the shock. The differential behavior of the rental rates of the two factors of production induces a substantial redistribution of income across classes of agents over the adjustment path. In fact, since the wage rate of an efficiency unit of labor drops, low-skilled workers' income and consumption decline. Also, because the economy attempts to reconstruct the (per-capita) capital stock, consumption of high-skilled drops more than that of low-skilled agents and rises above the pre-migration level after 16 years.

Overall, there are two highly intertwined effects following a migration of low-skilled workers. First, there is redistribution over the business cycle with holders of capital benefitting and workers being worse off as competition from the migrants decreases the wage rate. Second, there is a strong wealth effect in the long-run since high-skilled workers, which are now a smaller fraction of the population, permanently own more capital per head. This makes their income and consumption permanently higher and their hours permanently lower since the relative abundance of low-skilled labor, combined with their willingness to work more, results in substitution from high-skilled to low-skilled labor in production. Hence, unambiguously, migrations of low-skilled agents will be encouraged by those sectors of the economy which own capital and/or are highly productive.

The model of Ambler and Paquet (1994) with stochastic capital depreciation also generates responses which are qualitatively similar to ours over the business cycle. Two differences, however, can be noted. First, a temporary disturbance to the depreciation rate of capital does not change the steady-state of the economy so none of the longrun effects we describe are present in their paper. Second, our model in the perfectly substitutable case can also be interpreted as one with stochastic depreciation with the main difference being that now there is a stochastic adjustment cost to investment as well. Because such adjustment cost is stochastic and countercyclical, the responses of investment and output following a migration shock are smaller in magnitude and less persistent than those presented by Ambler and Paquet.

4.3 Basic Dynamics with imperfect substitutability

The dynamics of aggregate variables when labor is imperfect substitutable in production are not substantially altered. Quantitatively, only investment shows a sizable change with the impact increase now being smaller. As mentioned, with sufficiently low substitutability between the two types of labor, the relationship between hours in efficiency units and γ_t can change sign and this indeed happens here; long run output, investments and the capital stock now ends up being slightly higher than in the original steady state.

The major difference between perfect and imperfect substitutable labor is that now there is a wedge between high and low-skilled wages. The reason for this is clear: un-

4 THE DYNAMIC EFFECTS OF LABOR MIGRATION

less high-skilled agents increase their labor supply substantially, an inflow of low-skilled agents must increase the relative wage of high-skilled agents. Whether this also leads to an absolute increase in high-skilled wages depends on the exact value of, among other parameters, the elasticity of substitution between the two types of labor. In the case of Figure 2, the absolute wage of high-skilled wages still decrease on impact (but increase in the long-run). For lower elasticities high-skilled wages may increase on impact.

Over the business cycle, the wage gap between high and low-skilled agents increases and this produces an extra source of income transfers. The redistribution from low to high-skilled agents is substantial because it occurs in conjunction with the redistribution from labor to capital we have previously discussed, making low-skilled agents much worse off. Furthermore, such redistribution now takes place not only in the short-run but also in the long-run since low-skilled (high-skilled) wages will be permanently below (above) their initial level. Thus, low-skilled consumption decreases more than in the previous case while high-skilled consumption rises above the initial level much faster and increases more in the long-run.

In conclusion, while in the case of perfect substitution there is a redistribution of income between wage earners and capital owners, as the real rate increases while the wage rate declines, here there will also be redistribution between low and high-skilled wage earners. Notice that because wage effects dominate in magnitude relative changes in the price of capital and labor both over the business cycle and in the long-run, migrations of low-skilled agents will be strongly encouraged by high-skilled agents.

4.4 Migrations with low persistence

So far we have considered persistent migration flows, i.e. a positive inflow this year signal the presence of positive inflows next year. Next, we consider the case of migrations with low persistence ($\psi_e = 0.1$), which lead to a 2% long run change in the composition of the population. Because of the way we have designed the experiment, long run effects are identical to the previous ones and only the cyclical properties of the model will be affected. Also, comparing the outcomes of this experiment with the previous ones may suggest a rationale for choosing gradual vs. once and-for-all migration schemes. The results for this experiment are presented in Figure 3 for both the case $\rho = 0$ and $\rho = 0.25$.

In general, when persistence is low, the reaction of the economy will be reduced, since the informational content of the shock is smaller. In particular, in the case of perfect substitutable labor investment per-capita and hours in efficiency units will increase less so that output per-capita will decline more. In the case of imperfect substitutable labor, the wage gap between the two types of labor will be smaller both at impact and over the business cycle so that the effects we have described in the previous subsections will be attenuated. One interesting feature of the responses is that the investment increases more in the case of imperfect substitutes than in the case of perfect substitutes, while with more persistent shocks the opposite was true. In essence, the business cycle/redistribution consequences are magnified when migration flows are gradual since, in deciding the optimal response in the current period, agents take into account future migration developments.

4.5 Transient migration

We next consider transient migrations, in the sense that after the initial shock there will be a shock of similar or smaller magnitude in the opposite direction. Because of our modelling strategy, we can not consider the more interesting case of a temporary changes in the composition of the population (so that the initial and the final steady state share of high-skilled agents are the same). However, the chosen experiment represents the closest approximation we can construct using a linear AR representation for the shocks. We now model the migration process as:

$$e_t = \psi_e^1 e_{t-1} + \psi_e^2 e_{t-2} + \xi_t$$

and assume that $\psi_e^1 = -0.3$, $\psi_e^2 = 0.25$. Furthermore, the initial shock is assumed to be of the same size as in the baseline case. With the above process this reduces the long-run share of high-skilled agents from 75% to 74.24%.

Impulse responses for this experiment are presented in Figure 4. As expected, given that the compositional effect is smaller, long-run changes are smaller both in the perfect and the imperfect substitutable cases. By continuity argument, we conjecture that all long-run effects will disappear, if the share of skilled agents is constant in the long run. Over the business cycle the magnitude of the responses of aggregate and sectorial variables is smaller relative to the two baseline cases. This is to be expected as changes in the population (and in the share of skilled agents in the economy) this year are known to be followed by changes next year in the opposite direction.

Hence, when migrations are transient, they will induce less cyclical behavior in percapita aggregates and reduced long-run consequences on the receiving country.

4.6 Migrations of high-skilled workers

Finally, we turn attention to the case of migrants which are more skilled than the average native. This experiment is interesting because some countries (e.g. Australia or Canada) have adopted immigrations policies where the educational level of potential migrants is used as screening device to limit the flow of newcomers. Moreover, such an exercise may provide indications on whether this policy, which is clearly beneficial in the long-run, has also important positive effects over the business cycle.

To model high-skilled migrations we assume that the growth rate of low-skilled agents is now constant and let $g_t^s = (1+e_t)g^u$. This is not necessarily the symmetric equivalent of a drop in the low-skilled portion of the population since high-skilled migrants carry capital with them. To make the analysis tractable we assume (i) e_t is a iid shock and (ii) the percapita capital stock migrants carry is the same as the per-capita capital stock of native high-skilled workers. Had we not made these assumptions, we would end up having an unbound number of types of agents in the economy (distinguished by their asset holdings at each point in time) and we would not be able to analyze this case within the context of the model we are considering. Once these assumptions are made, the impulse responses are qualitatively the same as those we presented in figures 1 and 2 except that now there is a sign change in all the responses³.

Hence, an inflow of high-skilled agents increases (decrease) activity in the long run when the two types of labor are perfect substitutes (imperfect substitutes for low enough substitutability) substitutes. In the short-run such a migration will decrease investments and interest rates and increase the wage rate. Thus, low-skilled native will favor limiting migrat type to high-skilled agents who carry large amounts of capital with them.

5 Migrations and the Welfare State

In the previous section we have seen that a temporary inflow of low-skilled agents produces a redistribution of income from low-skilled natives to high-skilled/capital holders, making them substantially better off over the business cycle. It is therefore conceivable to think of transfer rules which attempts to limit the damages to low-skilled agents, while maintaining some of the gains for the high-skilled/capital holders. Here we consider the effects of migration disturbances when there is government which either passively or actively pursues redistributive policies. In all cases we analyze we assumed that the policy is exogeneous. This assumption is made in order to simplify the analysis but it would be feasible to endogenize it, for example through voting on redistribution policies. For our purposes, however, it seems natural to consider exogenous policies since we consider marginal changes in the population composition.

We consider three different schemes:

• a passive rule (\mathbf{PR}) in which the income tax rate is taken parametrically;

• an egalitarian rule (\mathbf{ER}) , where the income tax rate on high-skilled workers is chosen so as to keep the ratio of income of high to low-skilled agents constant;

• an insurance rule (IR) where the government insures the consumption of low-skilled agents from any type of fluctuations.

Under **PR**, the government sets $\tau_t^s = \tau^s$, $\forall t$, and varies μ_t period-by-period so as to satisfy the government budget constraint, (10). In the **ER**-system, it chooses the income tax rate on high-skilled agents (τ_t^s) and the transfer rate (μ_t) to satisfy its budget

³The most appropriate case to look at is the one with low migration persistence. The reason is that we assume the migrant, *at every point in time*, hold the same amount of capital as the native high-skilled agents. Thus, in principle, migrants are indexed by the state vector of the economy.

constraint and to target a given income difference between the two groups of agents, i.e.:

$$(1 - \tau_t^s) y_t^s = \eta \left(1 - (\tau_t^s - \mu_t) \right) y_t^u$$
(17)

where η is the wedge between the after-tax income of the two types of agents.

Under IR, taxes and transfers are chosen to satisfy the government budget constraint and to make:

$$(1 - (\tau_t^s - \mu_t)) y_t^u = \overline{y}^u \tag{18}$$

where \bar{y}^u is a constant.

The reason for studying these rules is simple. The first is chosen as benchmark: we would like to know how the responses of endogenous variables differ from those previously described when a government, without an explicit objective function, is present in the economy. The second rule is common both in theoretical studies examining the static effects of migration (see Razin and Sadka (1995)) and in the real world (see, e.g., Canada and the Scandinavian countries). Egalitarian rules also turn out to be sufficiently popular as redistributive tools in standard models of public finance (see e.g. Auerbach and Kotlikoff (1987)) to grant them a particular status in our study. The third rule has been designed keeping in mind both the structure of the model, where low-skilled agents are unable to insure income fluctuations, and some concerns of European policy makers, who foresee the provision of public insurance over the business cycle as a future task for a European fiscal unit (see e.g. Padoa-Schioppa (1987)). Note that the second rule allows for fluctuations in low-skilled workers' income over the business cycle as long as they are proportional to high-skilled income, while this is not permitted with the **IR** rule.

We assume that before the experiments take place the income tax rate on skilled agents is 5%, a number close to those observed in many countries once it is taken into account that the tax rate in our model relates only to the parts of the government budget associated with redistribution. To save space we concentrate on the case $\rho = 0$. Since when $\rho > 0$ the redistributive effects of a migration disturbance against low-skilled agents are exasperated, the qualitative features of the dynamics will be similar to those described here, but the quantitative changes would be more dramatic.

5.1 Parametric Tax Rule

The impulse responses obtained under the PR rule are presented in Figure 5.

Overall, there are no major qualitative changes in the responses of aggregate variables relative to the baseline case. The reason is that when τ_t^s is constant over the cycle high-skilled agents behave as in the case where the tax rate is zero (except for the change in the elasticities brought about by the associated change in the capital-output ratio). Hence, their decisions are unchanged relative to the baseline model.

The major difference relative to the impulse responses of Figure 1 is in the sectorial variables for low-skilled agents. Because the tax rate on high-skilled workers is constant

5 MIGRATIONS AND THE WELFARE STATE

and the number of low-skilled workers increases and the per low-skilled worker transfers decrease. This is clear from the dynamics of μ_t which not only declines, but it is negative, implying higher taxation for low-skilled. Hence low-skilled consumption declines over the business cycle making them worse off than in the baseline case.

Thus, this tax scheme is not able to insure low-skilled agents from the consequences of a migration of low-skilled workers. On the contrary, when a passive tax rule is in place, low-skilled workers are penalized in two ways by the existence of migrations. First, competition in the labor market causes a decline in their wage, income and consumption. Second, because the amount that high-skilled workers contribute to the welfare system shrinks and the share of low-skilled workers increases, low-skilled agents are taxed more heavily to make up for the loss of government revenue and the larger number of agents which are potentially recipients of the transfers.

5.2 Egalitarian Tax Rule

In this exercise we treat η as a parameter and the benchmark value corresponds to the value implied by a 5% income tax rate on high-skilled agents. Given other parameters, this implies that the low-killed after-tax income is 31% of the after tax income of the high-skilled before any shock takes place⁴. The impulse responses following a migration disturbance appear in Figure 6.

The most important change in per-capita variables relative to the baseline case occurs because the marginal tax rate on high-skilled agents increases substantially as their beforetax income increases. Higher income taxes discourage investments both on impact and over the business cycle and this leads to a more pronounced negative long-run effect on output, investment and the capital stock.

At the sectorial level, the changes relative to the two previous cases are substantial. Because an inflow of low-skilled migrants permanently alters the income distribution in favor of capital holders, the government will tax high-skilled workers/capital owners more heavily. This implies that their consumption decreases on impact and will be persistently lower over the business cycle. The higher tax rate on high-skilled agents also implies that the pool of revenues available for redistribution increases and this has beneficial effects on low-skilled consumption on impact. However, this favorable occurrence is partly due to the increase in high-skilled labor supply and over the business cycle low-skilled consumption decreases but by much less than in the previous case. Furthermore, the increase in taxes adversely affect high-skilled income and this limits the amount of redistribution that the welfare state can engage in. Low-skilled agents will therefore also bear higher tax rates

⁴We have also experimented with a set-up where η is a deterministic function of γ_t . This might be relevant in terms of political economy based determination of redistribution. The dynamics generated in this set-up are qualitatively similar to those obtained with parametric tax rules but the effects are magnified, particularly in the case of a migration shock. We do not report this case because such effects are likely to take place over longer horizons than those that we are mainly concerned with.

5 MIGRATIONS AND THE WELFARE STATE

during the adjustment to the new steady state and this combined with a lower wage implies that their consumption will be lower than in the two previous cases. Because the composition of the population has permanently changed, tax rates on both types of agents will permanently increase in the long run.

In sum, this redistribution scheme has important side effects in the presence of migrations. With inflow of low-skilled agents, the "size" of the welfare state increases and skilled agents, which are a smaller fraction of the population, contribute to a larger percentage of total government revenues. Because they are taxed more heavily they contract investments and this in turn generates a reduction of per-capita income that exceeds that observed in the two previous cases. Notice that this occurs despite the fact that the after tax real rate of interest is positive over the business cycle. Thus, this tax system effectively shifts the costs from low-skilled to high-skilled agents.

5.3 Insurance Tax Rule

The impulse responses of the variables following a migration disturbance are presented in Figure 7. In general, the aggregate dynamics following a low-skilled migration shock are similar to those obtained with an egalitarian tax rule.

The responses of sectorial variables are more dramatic because of the associated tax effects: an inflow of low-skilled agents leads to higher marginal tax rates on high-skilled agents and this effect is now stronger than before. Hence, investments will be discouraged and the negative effect on the long-run capital stock is large. This process is somewhat self-enforcing: tax rates must increase along the business cycle because of the negative effect that such an increase has on capital accumulation, and this leads to magnified effects both over the business cycle and in the long run.

By eliminating short run fluctuations in low-skilled consumption, this rule heavily penalizes high-skilled workers in the presence of unexpected migrations since low-skilled income declines while at the same time there is a larger number of people that needs to be insured. Compared with an egalitarian rule, the burden of the short run adjustment is more heavily tilted toward high-skilled agents which now bear up to an 18% increase in their marginal tax rate over the business cycle.

To summarize, when unexpected migrations of low-skilled agents occur, the size of the welfare state increases. When high-skilled agents' taxation is fixed over the business cycle, redistribution policies turn against low-skilled agents which are now worse off because there is a smaller pool of resources to be redistributed to a larger number of agents. When high-skilled workers contribute with varying amounts to the welfare state over the adjustment path, the cyclical consequences may be more dramatic as tax disincentives depress capital accumulation and the per-capita income. With egalitarian taxes both classes of agents are penalized over the business cycle relative to the case of no redistribution. With insurance taxes, the full burden of the increase in the welfare state falls on high-skilled agents and the depressing effects are significantly larger.

5.4 Illegal migrations

Although the evidence of section 2 concerns legal migration, it is well known that the magnitude of illegal migrants flows to some OECD regions is probably as large as the one of legal migrants. It is therefore interesting to consider what illegal migrations do in the context of our model when there is an active welfare state.

Given the simplicity of the economy, we capture only one aspect to the very complex problem of illegal migration, namely that illegal migrants enter the labor market, but are not entitled to the welfare benefits that the government provides to native low-skilled agents. In our setup, illegal migrants do not pay taxes either but we could easily amend this and have them pay payroll taxes without receiving any transfers. This experiment is interesting to policymakers since with the recent IIRIRA ACT of 1996, even *legal* migrant in the US will no longer be entitled to the welfare benefits enjoyed by natives.

Relative to the baseline case, the main change in the model⁵ is now that the government budget constraint reads:

$$\tau_t^s \gamma_0 y_t^s = (\mu_t - \tau_t^s)(1 - \gamma_0) y_t^u \tag{19}$$

where γ_0 and $1 - \gamma_0$ denote the pre-migration shares of the two types of native agents.

We present the impulse responses of this experiment in Figure 8 when there is an egalitarian welfare state. The qualitative behavior of per-capita variables is unchanged and even quantitatively differences are small relative to the case considered in Figure 6. However, at the disaggregate level, changes are important. Since the pool of potential users of the welfare state decreases relative to the case where low-skilled natives and migrants enjoy an egalitarian redistribution, high-skilled agents will be taxed at a lower rate and this allows them to increase their consumption along the business cycle (although, quantitatively, the change is small). On the other hand, since tax-revenues increase following the increase in high-skilled income, the welfare state redistributes more to native low-skilled agents, which now experience an increase in their after-tax incomes. Finally, migrant income, which is the same as native low-skilled income before redistribution declines on impact and stays below zero over the business cycle.

Hence, one way to reduce the negative effects on both capital accumulation and lowskilled consumption due the presence of an egalitarian government is to reduce (or eliminate) the benefits for the newcomers. Native low-skilled will favor this proposal and prefer illegal to legal migrants when there is an egalitarian state because they can reap larger per-capita benefits per unit of consumption subtracted to high-skilled agents. Native high-skilled will be better off because they pay lower income taxes. These gains, however, are entirely at the cost of the migrants who are excluded from the welfare state.

⁵This case also requires one to change the goods marget clearing condition because local low-skilled agents and the migrants have different after-tax incomes.

5.5 Restricting the labor market participation of migrants

We finally consider the case where migrants are not allowed to enter the labor market of the receiving country, but are compensated enjoying the same level redistribution of native low-skilled agents. This could have represented the situation of migrants who reached the US for family unification or with a refugee status prior to August 1996.

Since the migrants effectively have a consumption rate of 1, the experiment we consider is equivalent to a permanent increase in government spending financed through distortionary taxes (apart from the fact that migrants also increase the population of the country). We assume that transfers to the migrants are a fixed fraction of the native low-skilled agents' initial income $c_t^{IM} = \varphi y_0^u$ where y_0^u denotes low-skilled agents' premigration income and φ determines the level of the transfers to the migrants. We assume that $\varphi = 25\%$ and that the government engages in egalitarian redistribution.

Figure 9 illustrates the dynamic effects in this case. At aggregate level all variables now decline on impact and over the business cycle: particularly strong is the effect on investment which now declines by 1.5% while in figure 6 was increasing by about 1.5%. The reason is that the government needs to increase taxes on high-skilled to finance transfer payments to migrants and this occurs in a situation where there is no initial redistribution to high-skilled/capitalists since the wage and the rate of interest are unchanged.

The most significant effects are apparent at the disaggregated level. Since total resources decrease slightly and the government transfer income to migrants, consumption of both types of natives decrease. The percentage decreases are of approximately the same size because they are driven by tax effects and they become substantial over the adjustment path, as the percentage decreases in consumption of both types reaches 0.5%. For high-skilled agents this decrease is smaller than that observed in figure 6 while the opposite is the case for low-skilled agents. Thus with an egalitarian welfare state, highskilled agents would oppose admitting migrants to the labor market while low-skilled agents would prefer they do enter.

This outcome might seem counterintuitive: restricting labor market participation of migrants avoids the competition that harms native low-skilled agents. The explanation can be found in the specification of the welfare state. Under the egalitarian system the government keeps the wedge between high-skilled and low-skilled after-tax incomes fixed. Thus, when migrants *are* admitted to the labor market the welfare state engages in "more" redistribution because high-skilled agents' income rises as the return on capital increases. Therefore, this redistribution makes local low-skilled agents better off relative to the case considered here.

It is worth considering whether this result crucially depends on the level of transfers that the immigrants receive and the details of the welfare transfer. First, whether high-skilled agents would prefer not to admit immigrants to the labor market clearly depends on the tax burden. If φ is increased to 50% we find that even high-skilled agents would favor admitting migrants to the labor market. Second, if the welfare state operates under the

IR system native low-skilled agents are equally well off whether immigrants are admitted to the labor market or not. Thus, in this case the question is whether the high-skilled agents are better off with or without admitting the migrants to the labor market. Again, this depends on the transfers parameter, φ : for values above 30%, high-skilled agents would prefer that the migrants enter the labor market.

In summary, whether natives prefer to have migrants in the labor market or not depends on the level of transfers and on the type of natives we consider. In general, for low transfers high-skilled agents would prefer to keep migrants out of the labor market while low-skilled agents would favor their labor market participation.

Finally it is worth mentioning that, as far as the welfare of natives is concerned, this experiment resembles a foreign-aid program in which the government transfers funds to a foreign country raising taxes on natives. Clearly, such a program would not affect the per-capita variables in the same way (since the domestic population is unchanged in the case of foreign-aid) but the qualitative effects on natives would be quite similar.

6 Welfare Gains and Losses From Migrations

The analysis so far has demonstrated that migrations of low-skilled agents may have very different implications for the receiving country depending on the type of redistributive policy pursued by the government. Without taxes, skilled agents/capital holders benefit from migrations and this may explain why there was not much public outcry against migrations in the 50's and the 60's. However with redistributive taxes, the burden of migration costs is shifted in part or completely onto high skilled workers and capital owners.

In this section, we ask: how much are residents willing to pay in terms of consumption in order to keep migrants out of the country? In other words, we ask how much a country (or sectors of a country) is willing to devote to programs which would restrict the flow of migrants in exchange of foreign aids programs to nations with a potential pool of migrants. This question is not purely academic. Several countries, e.g., Italy in the case of Albanians, have engaged in active foreign aid policies to curb massive inflows of migrants responding to political and economic changes occurring in their countries. Others, e.g. the Scandinavian countries, where well developed welfare states are in place, might find themselves burdened by the potential costs of admitting migrants into the country. The US, on the other hand, has diverted funds from welfare provisions of migrants to programs controlling borders easily penetrable by migrants.

We compute long-run welfare measures along the lines of Lucas (1987). In particular, we compute the percentage change in consumption such that the utility of the native population is the same before and after the migration of low-skilled workers. It is possible also to evaluate the welfare effects taking the dynamic adjustment path into account, but we choose to abstract from this in order to get a welfare metric that is more easily interpretable. Because native population is a heterogeneous group, we compute two such measures. The first imputes all the costs or gains to skilled workers/capitalists. The other measure assumes that the two different groups of native households share the costs or gains and weights the two groups by their initial share of the population in the social welfare function. This gives us the following two welfare measures:

$$\lambda_1 = 1 - \frac{c_1^s}{c_0^s} \left(\frac{1 - h_1^s}{1 - h_0^s} \right)^{(1-\theta)/\theta}$$
(20)

$$\lambda_2 = 1 - \left(\frac{W_0}{W_1}\right)^{1/\theta(1-\sigma)}$$
(21)

where $c_0^s(c_1^s)$ denotes high-skilled consumption prior to (after) the inflow of low-skilled workers, $h_0^s(h_1^s)$ are hours worked prior to (after) the inflow and

$$W_0 \equiv \gamma_0 \left(c_0^s\right)^{\theta(1-\sigma)} \left(1 - h_0^s\right)^{(1-\theta)(1-\sigma)} + \left(1 - \gamma_0\right) \left(c_0^u\right)^{\theta(1-\sigma)} \left(1 - h_0^s\right)^{(1-\theta)(1-\sigma)}$$
(22)

$$W_1 \equiv \gamma_0 \left(c_1^s \right)^{\theta(1-\sigma)} \left(1 - h_1^s \right)^{(1-\theta)(1-\sigma)} + \left(1 - \gamma_0 \right) \left(c_1^u \right)^{\theta(1-\sigma)} \left(1 - h_1^u \right)^{(1-\theta)(1-\sigma)}$$
(23)

We compute these two measures for: (i) zero taxes, (ii) a constant income tax rate equal to 5 percent, (iii) an egalitarian tax-system, and (iv) an insurance tax-system. In all cases we assume that $\gamma_0 = 0.75$ and then compute λ 's for 1 to 5 percent changes in the composition of the population. For each of these cases we look at the welfare effects for three different relative productivities of the two groups of households. In one case we let both groups have exactly the same productivities so that the only difference is due to the ownership of the capital stock. In the second case we set $\omega = 1.3$, and the final case corresponds to the parameterization of Kydland (1984), i.e. $\omega = 2.0$. As ω increases, the productivity difference between the average natives and the migrants increases. Also, we examine two different values of the elasticity of substitution parameter, $\rho = 0.0$ and $\rho = 0.25$.

Before discussing the results, it is worth stressing that our calculations may be affected by several mechanisms which we have not modelled in the paper. In particular, we have assumed that the skill differences are exogenous. It has been observed in many studies that second generation immigrants do not differ significantly from natives in terms of skills. However, it is also true (see Felderer (1994)) that human capital accumulation of migrants is very slow. While we believe that skill acquisition is an important issue one needs to model before taking a firm view of the welfare effects of migrations, we also believe that, as a first approximation, it may be interesting to compute welfare costs disregarding this issue (Storesletten (1995) considers a model with these effects).

Table 2 reports welfare measures for $\rho = 0$ while table 3 welfare measures for $\rho = 0.25$ Part A of Table 2 examines the case where there is no redistribution. It is quite clear that in this case migrations lead to an increase in both welfare measures. Furthermore, the welfare gain is increasing in the size of the migration flow and decreasing in the productivity differential. The intuition is straightforward: an inflow of labor unambiguously leads to a welfare gain for the capitalists. To realize this, one can simply consider the welfare effects if the capital stock was kept constant. In that case the capitalists would enjoy a higher income due to the higher return on capital because the capital-labor ratio decreases. This is the case investigated by Berry and Soligo (1969), Borjas (1994a,b) in a model with homogenous natives, and Benhabib (1996) in a model with heterogeneous agents. However, here there is an incentive to accumulate capital and the economy ends up with an unchanged capital-output ratio (but the capitalists hold per-household more capital in the new steady-state), while interest rates and wages are unchanged. Hence, capital-owners will unambiguously be better off while the utility of low-skilled agents is unchanged. When the productivity differences increase, the corresponding change in the initial effective capital-labor ratio is smaller and less new capital will be accumulated. Clearly, welfare gains are smaller when the two groups share the benefits.

Panel B of table 2 corresponds to a 5 percent income tax which we hold constant while the effective redistribution to the low-skilled agents is varied so that the government budget constraint is satisfied. When high-skilled agents bear the costs, welfare costs are exactly the same as in the previous case since the only distortion associated with the income tax is to lower output through its' effect on the capital-output ratio. Hours worked - and hence the labor input in efficiency units - are unaffected because of the preference specification we use. Hence, in percentage terms, the welfare gains for high-skilled are exactly the same as with a zero tax rate. The effects on the low-skilled workers, however, change. With a constant income tax rate, an inflow of low-skilled workers implies that there are fewer resources available for redistribution. Hence high-skilled workers cannot compensate low-skilled agents for this loss and aggregate welfare declines. Note again, that welfare losses are increasing in ω .

Part C of Table 2 reports the results when an egalitarian rule is in place. The results differ from the previous two cases because the marginal income tax rate changes. It is important to recall that in this case the ratio of the after tax income levels of the two groups is kept constant and equal to η . The inflow of low-skilled labor increases the before tax income of high-skilled agents and taxes are increased to keep η constant. This implies that steady-state capital-output ratio declines, that there is a welfare loss for high-skilled workers, which again is increasing in ω and, in contrast to above, that the welfare cost is decreasing in ω since the tax effect is smaller the more productive high-skilled workers are. Note that because low-skilled agents benefit from this program, the economy wide losses are smaller than those suffered by high-skilled agents for each ω .

With an insurance system (Part D), the welfare effects of migrations are very similar. With an inflow of migrants taxes need to be increased in order to maintain the after tax income that the welfare state promised to the low-skilled. When the productivity differences are sufficiently small, the costs for the capital owners are small, but they increase with ω . Note also that in this latter case costs are significant for high-skilled workers since tax rates are slightly higher than those experienced with an egalitarian rule while low-skilled agents welfare is unchanged.

When $\rho = 0.25$ the qualitative the effects are similar to those presented in table 2.

7 SUMMARY AND CONCLUSIONS

The only significant difference is in the case $\omega = 2.0$ for the case of no taxes since now there are welfare losses while before there were gains. This is due to the fact that when the productivity parameter is high, the differential effect on wages is large leading to a larger losses for the low-skilled component of the population. Two other facts need to be noticed. First, tax rates with ER and IR rules are much larger than in the previous case. The reason is that the before tax income differential between the two types of agents is much larger here. Second, the magnitude of the gains and losses in the four different schemes is magnified. For example, high-skilled agents would be willing to give up 1.7% of annual consumption to admit a flow of low-skilled migrants which change the composition of population by 1% with no taxes but they are willing to pay up to 0.5% of their consumption when there are insurance taxes to keep the same amount of migrants away. Once again this magnification is due to the presence of two redistributive margins and that taxes on high-skilled are very high.

To sum up, the quantitative effects of low-skilled labor migration crucially depend on the size of the immigration flow, the relative productivity difference between migrants and natives and on the redistribution system in place in the receiving country. When the receiving country does not engage in redistribution, migrations are unambiguously welfare improving for high-skilled agents and the receiving country, notwithstanding the possible short-run costs associated with adjustments of the capital stock, should encourage migration of this type of agents. However, if the receiving country does engage in redistribution, as is common in most OECD countries, welfare gains are turned into huge costs, which may permanently affect the long run prospects of the economy.

7 Summary and Conclusions

In this paper we analyzed the quantitative effects of low-skilled migrations in a dynamic model with capital accumulation. Migrations to OECD countries have been substantial over a prolonged period of time and it seems natural to investigate the macroeconomic consequences of such flows over the business cycle. One characteristic emphasized in the demographic literature is that migrants to OECD countries are less skilled than the average native worker and this aspect was included in our analysis.

We find that an inflow of low-skilled workers induces adjustments which resemble those taking place if the capital stock were suddenly disrupted. Such a change leads to higher investments over the business cycle, and if all agents were identical, this would be the only macroeconomic adjustment taking place. However, with an heterogeneous workforce, an inflow of low-skilled workers has also a composition effect which leads both to long-run changes and to differential impact on high and low-skilled agents. In the long-run migrations lead to a decrease in per-capita capital, output and investment if the two types of workers are perfect substitute in production. Skilled agents are better off when there is a migration of low-skilled agents because they own the capital stock: in the short run, i.e. for a fixed capital stock, the return on capital increases, while in the

7 SUMMARY AND CONCLUSIONS

long run they have higher capital income because they each hold more capital per-capita. Low-skilled agents are worse off in the short run because wages fall following a migration flow but are unaffected in the long run as wages return to the original steady state. When the two types of agents are imperfect substitute a differential wage effect is added which sharpens the income gap between high and low-skilled agents both over the business cycle and the long run.

We analyze the sensitivity of the results to alterations of the persistence of the shocks, the size of the long run compositional effects and the relative skill differential of migrants and natives and found some rationale for policies preferring transient to permanent migrations and high-skilled to low-skilled migrations over the business cycle.

Our conclusions are altered when there is redistribution among classes of agents. First, we consider a case where the marginal income tax rate is kept constant across classes of agents while government revenues are redistributed to low-skilled agents. With this scheme low-skilled agents are worse off in the long run because there are more agents to share a fixed pool of government revenues. Second, we consider an egalitarian scheme in which the government keeps the ratio of high to low-skilled agents after tax income constant. In this case, there is a negative effect on the capital stock since a flow of low-skilled agents forces high-skilled agents to finance a larger welfare state. The final case we considered was one in which the government insures low-skilled consumption over the business cycle and adjusts the tax rate accordingly. Here, an inflow of low-skilled workers leads to a prolonged period of adjustment and to an overall decline in welfare for both types of agents.

As a corollary to our analysis, we consider the effects of an inflow of illegal migrants and of restrictions on the labor market activities of migrants. In our set-up the distinguishing feature of illegal migrations is that these agents do not receive transfers and do not pay income taxes. When the state is egalitarian we find, maybe surprisingly, that both types of agents encourage illegal immigration! The reason for this is that low-skilled agents are net-receivers of transfers from the welfare state. Thus, excluding these migrants from the welfare state means that taxes increase less after an inflow of low-skilled agents, taxes needs to be increased less, which is favored by the high-skilled agents, and that the tax revenues are redistributed to the native low-skilled agents only. Another surprising result is that, when these is a welfare state in place, low-skilled natives favor the admittance of migrants to the labor market while skilled agents will oppose it. The reason is that when the migrants are allowed to work the welfare state will be more prone to redistribute income, which harms high-skilled agents and improve the welfare of low-skilled agents.

It is important to stress that our analysis constitutes only a first look at the problem and it would probably be important to endogenize skill formation before we can give a definite answer to the questions we have posed in the introduction. Empirical evidence suggests that initial skill differences between natives and migrants disappear over time, although at a very slow rate, and this may modify the results of our welfare calculations. Another issue that one may wish to address is whether international repercussions are

7 SUMMARY AND CONCLUSIONS

important and whether or not international aid schemes lead to improved conditions in international settings. These issues will be taken up in future research.

References

Ambler, A. and Paquet, A., 1994, "Stochastic Depreciation and the Business Cycle", International Economic Review, 35, 101-116

Auerbach, A.J. and L.J. Kotlikoff, 1987. *Dynamic Fiscal Policy*, Cambridge: Cambridge University Press.

Bean, C. and Pissarrides, C., 1991, "Skill Shortage and Structural Unemployment in Britain: a (Mis)match Approach" in F. Padoa Schioppa, (ed.) *Mismatch and Labor Mobility*, Cambridge: Cambridge University Press.

Benhabib, J., 1996. On the Political Economy of Immigration, "European Economic Review".

Berry, R.A. and R. Soligo, 1969. "Some Welfare Aspects of International Migration", *Journal of Political Economy*, 77, 784-94.

Borjas, G., 1994a, "The Economics of Immigration", Journal of Economic Literature, XXXII, 1667-1717.

Borjas, G., 1994b, "The Economic Benefits from Immigration". Journal of Economic Perspectives, 9(2), 3-22.

Burnside, Craig, Martin Eichenbaum, and Sergio Rebelo, 1993, "Labor Hoarding and the Business Cycle". Journal of Political Economy 101(2), 245-73.

Campbell, J.Y. and N. Gregory Mankiw, 1989. "Consumption, Income and Interest Rates - Reinterpreting the Time Series Evidence", *NBER Macroeconomics Annual*, 4, 185-216.

Felderer, B., 1994, "Immigration, the Labor Market and Structural Adjustments: the case of Germany", in H. Siebert (ed.), *Migrations a Challenge for Europe, Symposium 1993*, Institut fur Weltwirtschaft an der Universitat Kiel, J.C.B. Morh, Tubingen.

Lucas, R., 1987, Models of Business Cycles, Basic Blackwell.

Kydland, Finn E., 1984, "Labor-Force Heterogeneity and the Business Cycle", Carnegie-Rochester Conference Series on Public Policy 21, 173-208.

References

Kydland, Finn E., 1995. "Business Cycles and Aggregate Labor Market Fluctuations", in T. Cooley, (ed.), *Frontiers of Business Cycle Research*, Princeton University Press, Princeton, NJ.

Mankiw, N. Gregory and Stephen P. Zeldes, 1991. "The Consumption of Stockholders and Non-Stockholders", *Journal of Financial Economics* 29, 97-112.

OECD,1994, Trends in International Migrations, SOPEMI.

Padoa-Schioppa, T. 1987, Efficiency, Stability and Equity. Oxford: Oxford University Press.

Razin, A. and Sadka, E., 1993, "Interaction between International Migration and International Trade: Positive and Normative Aspects". Manuscript, Tel Aviv University.

Razin, A. and Sadka, E. 1995, "Resisting Migration: Wage Rigidity and Income Distribution". *American Economic Review*, 85(2), 312-16.

Rios-Rull, V. 1993, "Working in the Market, Working at Home and the Acquisition of Skills: A General Equilibrium Approach". *American Economic Review*, 83, 893-907.

Schrettl, W., 1992, "Transition with Insurance: German Unification Reconsidered", Oxford Review of Economic Policy, 8, 144-155.

Storesletten, K., 1995, "Sustaining Fiscal Policy Through Immigration", unpublished manuscript, Stockholm University.

Super, D. and Daskal, J., 1997, "States and Legal Immigrants: Does a Block Grant Make Sense?", Center on Budget and Policy Promotion, downloaded from http://www.cbpp.org.

Zimmerman, K., 1995, "Tackling the European Migration Problem", Journal of Economic Perspectives, 9(2), 45-62.

Appendix A: Steady-State Effects

In this appendix we discuss the steady-state effects of changes in the share of skilled agents. To start with, we collect various steady-state conditions and we will then analyze the steady-state changes in steps of increasing generality. The model delivers the following steady-state conditions (below all variables are evaluated at the non-stochastic steady-state):

$$\frac{1-\theta}{\theta}\frac{c^s}{1-h^s} = (1-\tau^s)w^s \tag{24}$$

$$c^u = (1 - \tau^u) \theta w^u \tag{25}$$

$$w^{s} = \alpha \left(\frac{y}{h^{e}}\right) \omega \left(\gamma h^{s}\right)^{-\rho} \tag{26}$$

$$w^{u} = \alpha \left(\frac{y}{h^{e}}\right) \left(\left(1 - \gamma\right)\theta\right)^{-\rho} \tag{27}$$

$$y = \gamma c^s + (1 - \gamma) c^u + x \tag{28}$$

$$y = z^{1/\alpha} \left(\frac{k}{y}\right)^{(1-\alpha)/\alpha} (h^e)^{1/(1-\rho)}$$
(29)

$$\frac{k}{y} = \frac{g^p \left(1 - \alpha\right)}{r} \tag{30}$$

$$r = \frac{1 - \beta \left(1 - \delta\right)}{\beta \left(1 - \tau^s\right)} \tag{31}$$

$$s_x = \frac{x}{y} = (1 - (1 - \delta) / g^p) \frac{k}{y}$$
(32)

$$\gamma \tau^s y^s = -(1-\gamma) \tau^u y^u \tag{33}$$

$$h^{e} \equiv \omega \left(\gamma h^{s}\right)^{1-\rho} + \left(\left(1-\gamma\right)\theta\right)^{1-\rho} \tag{34}$$

These equations will together with a tax-rule determine the steady-state of the economy. In order to analyze the issues of interest it is convenient to start by setting all taxes to zero.

A.1 No Taxes

In this case it follows from (30) and (31) that the capital-output ratio depends only on preferences and technology:

$$\frac{k}{y} = \frac{\beta g^p \left(1 - \alpha\right)}{1 - \beta \left(1 - \delta\right)} \tag{35}$$

Thus, changes in the share of the two types of agents will not affect the capital-output ratio. Notice also that the output share of investment is independent of γ . From equation (29) it follows that the effects on aggregate activity are solely determined by the effect on h^e , which, in turn, is depends on skilled hours, the productivity parameter ω , and the shares of the two Appendix A

types of agents. Because skilled hours is the only endogenous variable, we concentrate initially on the determination of h^s .

Combining (24) with (26) and (25) with (27) gives us:

$$\frac{1-\theta}{\theta}\frac{c^s}{1-h^s} = \alpha \left(\frac{y}{h^e}\right)\omega \left(\gamma h^s\right)^{-\rho} \tag{36}$$

$$c^{u} = \alpha \left(\frac{y}{h^{e}}\right) \left(\left(1 - \gamma\right)\theta\right)^{-\rho} \tag{37}$$

Inserting into (28) and rearranging gives us:

$$(1 - s_x - \alpha) \left(1 + \frac{1}{\omega} \left(\frac{(1 - \gamma)\theta}{\gamma h^s} \right)^{1 - \rho} \right) = \alpha \frac{\theta - h^s}{(1 - \theta) h^s}$$
(38)

This relationship defines h^s as an implicit function of γ . Note that since $s_x + \alpha < 1$, we have that $h^s < \theta = h^u$. There are two special cases where we can explicitly solve (38) for h^s .

A.1.1 Perfect Substitutes

When the two types of labor are perfect substitutes, i.e. when $\rho = 0$, (38) and (34) can easily be solved for h^s and h^e :

$$h^{s} = \frac{\alpha\theta - (1 - s_{x} - \alpha)\frac{1}{\omega}\frac{1 - \gamma}{\gamma}\theta(1 - \theta)}{\alpha + (1 - s_{x} - \alpha)(1 - \theta)}$$
(39)

$$h^{e} = \frac{\alpha \theta \left((1 - \gamma) + \omega \gamma \right)}{\alpha + (1 - \theta) \left(1 - s_{x} - \alpha \right)}$$

$$\tag{40}$$

It follows immediately that:

$$\frac{\partial h^s}{\partial \gamma} = \frac{\left(1 - s_x - \alpha\right) \frac{\theta}{\omega} \frac{1}{\gamma^2}}{\alpha + \left(1 - s_x - \alpha\right) \left(1 - \theta\right)}$$

and that

$$\frac{\partial h^e}{\partial \gamma} = \frac{\alpha \theta}{\alpha + (1 - \theta) \left(1 - s_x - \alpha\right)} \left[\omega - 1\right]$$

Thus, an inflow of low-skilled labor decreases skilled hours and "effective" hours for $\omega > 1$.

A.1.2 Unitary Elasticity of Substitution

The second special case where one can easily solve for h^s is when $\rho = 1$. In this case (38) reduces to:

$$h^{s} = \frac{\alpha \theta \frac{\omega}{\omega+1}}{(1 - s_{x} - \alpha)(1 - \theta) + \alpha}$$

$$\tag{41}$$

The important thing to notice is that h^s is independent of γ . Thus, in this case, the effects of changes in the skill composition on aggregate activity is determined completely by the composition effect on $(h^e)^{1-\rho}$, see (29) and (34). It follows immediately that:

$$sign\left(\frac{\partial (h^e)^{1-\rho}}{\partial \gamma}\right) = sign\left(\frac{\omega}{1+\omega} - \gamma\right)$$

Thus, for large enough skill differences, or big enough shares of low-skilled agents, an inflow of low-skilled agents will increase aggregate activity and vice versa for low skill differences and high share of high-skilled agents.

A.1.3 The General Case

In the general case we cannot derive closed form solutions. We can, however, analyze the properties of skilled hours viewing (38) as a function implicitly defining h^s as a function of γ . The right hand side of (38) does not involve γ and therefore defines a function $f(h^s)$. f is a decreasing function of h^s . Notice also that $f(\theta) = 0$ and that $\lim_{h^s \to 0_+} f(h^s) = \infty$. The left hand side of (38) involves γ and we let this function be denoted by $g(h^s, \gamma)$. The derivative of this function with respect to h^s is given by:

$$\frac{\partial g\left(h^{s},\gamma\right)}{\partial h^{s}} = \left(\rho-1\right)\left(1-s_{x}-\alpha\right)\frac{1}{\omega}\left(\frac{\left(1-\gamma\right)\theta}{\gamma}\right)^{1-\rho}\left(h^{s}\right)^{\rho}$$

Thus, the left hand side of (38) is an increasing function of h^s for $\rho < 1$ and a decreasing function of h^s for $\rho > 1$. Notice also that the slope tends to zero for h^s tending to 0 for any $\rho \ge 0$, that $\lim_{h^s \to 0_+} g(h^s, \gamma) = l < \infty$, and that $g(\theta, \gamma) > 0$ for any $\gamma > 0$.

Thus, f and g will have a unique intersection for $h^s \in]0, \theta[$ and at this equilibrium g will cut f from below. This implies that the response of skilled hours to changes in the skill composition can be determined from the changes in $g(h^s, \gamma)$ due to γ . The partial derivative of g with respect to γ is given by:

$$\frac{\partial g\left(h^{s},\gamma\right)}{\partial\gamma} = \left(\rho-1\right)\left(1-s_{x}-\alpha\right)\frac{1}{\omega}\left(\frac{1-\gamma}{\gamma}\right)^{-\rho}\left(\frac{\theta}{h^{s}}\right)^{1-\rho}\frac{1}{\gamma^{2}}$$

This implies that h^s is an increasing function of γ for $\rho < 1$ and a decreasing function of γ for $\rho > 1$. An inflow of low-skilled agents will decrease skilled hours in the new steady state if the two types of labor are relatively good substitutes (i.e. if $\rho < 1$) and increase skilled labor supply if high-skilled and low-skilled labor are relatively poor substitutes.

Thus, the properties of h^s depend on whether the elasticity of substitution is greater or smaller than one. The effect on h^s does not translate directly into what happens with aggregate activity since this is determined by the effect on hours in efficiency units. The derivative of $(h^e)^{1-\rho}$ is given as:

$$\frac{\partial (h^e)^{1-\rho}}{\partial \gamma} = (h^e)^{-\rho} \left[\omega (h^s)^{-\rho} \gamma^{-\rho} \left(h^s + \gamma \frac{\partial h^s}{\partial \gamma} \right) - \theta^{1-\rho} (1-\gamma)^{-\rho} \right]$$
(42)

In general, this expression cannot be signed. We can, however, say the following. First, regardless of the elasticity of substitution, when γ is sufficiently high, the derivative will be negative since the last term in the square bracket will dominate for $\forall \rho > 0$. This effect is stronger the smaller is the elasticity of substitution. Thus, a high-share of high-skilled agents together with a low elasticity of substitution between the two types of labor means that an inflow of low-skilled agents leads to an increase in aggregate activity through the effect on hours in efficiency units. Note that this effect is somewhat mechanical since it comes because the marginal product of low-skilled labor is high when the share of these agents is low.

When γ is sufficiently low that the above effect does not dominate, the steady-state results crucially depend on the elasticity of substitution. If the elasticity is high, we know from above that $\partial h^s / \partial \gamma > 0$. Thus, it is especially likely in this case that an inflow of unskilled agents will lead to a decrease in activity and the likelihood of this event is higher the bigger is ω . If the elasticity of substitution is low, the effect on activity of a decrease in γ is more likely to be positive but the sign depends critically on other parameters.

A.2 Taxes

The addition of taxes creates additional problems in signing the derivatives and do not allow us to deliver many straight insights. We can, however, note the following two principles:

(a) The aggregate distortion created by the welfare state is caused solely by changes in the capital income tax rate, (b) hours worked are not distorted by marginal taxes.

To see this, it is convenient to start with the second principle which follows because of homothetic preferences.

It is particularly easy to show this property for low-skilled agents.

At the optimum the marginal rate of substitution (between consumption and leisure) is equalized to the relative price, i.e.:

$$\frac{1-\theta}{\theta} \frac{c_t^u}{T-h_t^u} = (1-\tau_t^u) \, w_t^u$$

Inserting the budget constraint then gives that:

$$\frac{h_t^u}{T - h_t^u} = \frac{\theta}{1 - \theta}$$

The reason for this result is that, because of homothetic preferences, the income and substitution effects of changes in the wage rate or in marginal taxes cancel out. This result holds at any point in time so that labor supply is not even adjusted to temporary changes in taxes or wages.

A similar result to above can be derived for high-skilled agents but in this case it only holds at the steady-state since intertemporal substitution triggers off adjustments over time in consumption and labor supply for dynamic changes in wages and marginal tax rates. With taxes, the equations equivalent to (35)-(37) reads: Appendix A

$$\frac{k}{y} = (1 - \tau^s) \frac{\beta g^p (1 - \alpha)}{1 - \beta (1 - \delta)}$$
(43)

$$\frac{1-\theta}{\theta}\frac{c^s}{1-h^s} = (1-\tau^s)\,\alpha\left(\frac{y}{h^e}\right)\omega\left(\gamma h^s\right)^{-\rho} \tag{44}$$

$$c^{u} = (1 - \tau^{s}) \alpha \left(\frac{y}{h^{e}}\right) \left((1 - \gamma) \theta\right)^{-\rho} + \frac{\tau^{s}}{1 - \gamma}$$

$$\tag{45}$$

Inserting into (28) and rearranging gives us:

$$(1 - \tau^{s}) - (1 - \tau^{s}) \frac{\beta g^{p} (1 - \alpha)}{1 - \beta (1 - \delta)} (1 - (1 - \delta) / g^{p})$$
(46)

$$= \alpha \frac{\theta - h^s}{(1-\theta) h^s} \frac{\omega \left(\gamma h^s\right)^{1-\rho}}{h^e} \left(1 - \tau^s\right) + \alpha \left(1 - \tau^s\right)$$
(47)

This relationship is independent of $(1 - \tau^s)$ and involves h^s as the only endogenous variable. Hence, high-skilled hours in the steady-state does not depend on the marginal income tax rate and this establishes the result.

Thus, we have established that steady state hours worked of both types of labor is neutral with respect to changes in the marginal tax rate. By inspecting equation (29) one can see that any extra steady-state effect associated with the presence of taxes is caused by changes in the capital-output ratio. From equation (43) we see that the capital-output ratio is determined by preferences, technology, and the marginal income tax rate levied on the skilled agents.

This implies that when τ^s is held constant over the adjustment path, the presence of a welfare state will not induce changes in the steady-state effects identified in the previous section. Under the two alternative redistribution schemes where τ^s is adjusted following a migration, any changes in this tax rate will be associated with further steady-state effects because of the distortionary effect on capital accumulation.

Appendix B

In Figure B.1 we illustrate how the economy adjusts following an unexpected 1% increase in productivity. We have assumed that the persistence of the productivity shock is 0.81, a value consistent with a quarterly value of 0.95, that the income tax rate is 0 and that $\rho = 0$. The figure reports the percentage deviation of each variable from its original steady-state and the notation should be self-explanatory. The presence of two types of workers does not modify the well-known dynamic pattern of adjustment of per-capita aggregate variables that these types of models generate in response to technological disturbances. On impact, investment increases to approximately 3.5% above its steady-state value. Output, hours and consumption also increase but with considerably smaller elasticities. Furthermore, since agents are risk averse, the consumption path is smoother than that of other variables. The capital stock also increases and peaks with a lag of 6-7 years. The aggregate marginal product of labor increases and then declines below the steady-state.

At disaggregated level, income of high-skilled agents grows more than that of low-skilled at impact because they are more productive and own the capital stock. Given the structure of the model, the consumption path of high-skilled agents is smoother than the one of low skilled agents as we have forced the latter agents to consume their income on a period-by-period basis. Also, the increase in total hours we observe is entirely due to high-skilled agents working harder since the labor supply of low-skilled agents is fixed. Since there are only $\gamma\%$ of high skilled agents in the population, the increase in their hours worked exceeds the increase in per-capita hours in the economy.

To summarize, when a favorable technology disturbance hits the economy, high-skilled workers provide the additional hours necessary to take advantage of the improved technological conditions and smooth over time the benefits of the disturbance by varying the level of investments. Low-skilled workers benefit from the improved technological conditions because their wages increase over the adjustment path. Note that, because low-skilled workers work a fix number of hours, we can think of this class of agents as being hoarded by the firms which vary the number of hours of high-skilled workers to meet their production requirements over the cycle. In this way, our economy resembles the one analyzed by Burnside, Eichenbaum and Rebelo (1993).

	A. Model With						
	$\omega = 1$.0	$\omega = 1$.3	$\omega = 2.0$		
γ	λ_0	λ_1	λ_0	λ_1	λ_0	λ_1	
0.74	-0.25	-0.16	-0.19	-0.13	-0.12	-0.08	
0.73	-0.50	-0.34	-0.39	-0.26	-0.25	-0.16	
0.72	-0.76	-0.52	-0.59	-0.39	-0.38	-0.25	
0.71	-1.03	-0.71	-0.79	-0.53	-0.52	-0.33	
0.70	-1.31	-0.90	-1.01	-0.68	-0.66	-0.42	

Table 2. Welfare Consequences of Migrations ($\rho = 0.0$)A. Model Without Taxes

B. Model With Fixed Income Tax

	$\omega = 1.0$		$\omega = 1$.3	$\omega = 2.0$		
γ	λ_0	λ_1	λ_0	λ_1	λ_0	λ_1	
0.74	-0.25	0.09	-0.19	0.21	-0.12	0.41	
0.73	-0.50	0.15	-0.39	0.39	-0.25	0.79	
0.72	-0.76	0.20	-0.59	0.54	-0.38	1.15	
0.71	-1.03	0.23	-0.79	0.68	-0.52	1.47	
0.70	-1.31	0.24	-1.01	0.80	-0.66	1.77	

C. Model With Egalitarian Tax-System

	$\omega =$	$\omega = 1.0$			$\omega = 1.3$			$\omega = 2.0$		
γ	τ	λ_0	λ_1	τ	λ_0	λ_1	au	λ_0	λ_1	
0.74	5.3	0.23	0.17	5.3	0.26	0.20	5.3	0.29	0.25	
0.73	5.6	0.48	0.34	5.6	0.52	0.41	5.5	0.59	0.51	
0.72	5.9	0.73	0.53	5.8	0.79	0.62	5.8	0.90	0.77	
0.71	6.2	0.99	0.72	6.1	1.07	0.84	6.1	1.21	1.04	
0.70	6.6	1.26	0.92	6.4	1.35	1.06	6.3	1.53	1.32	

	$\omega = 1.0$			$\omega = 1.3$			$\omega = 2.0$		
γ	au	λ_0	λ_1	τ	λ_0	λ_1	τ	λ_0	λ_1
0.74	5.3	0.24	0.17	5.3	0.29	0.20	5.3	0.35	0.23
0.73	5.6	0.49	0.35	5.6	0.59	0.41	5.6	0.71	0.48
0.72	5.9	0.75	0.53	5.9	0.90	0.63	5.9	1.09	0.73
0.71	6.3	1.03	0.73	6.2	1.23	0.86	6.2	1.47	0.99
0.70	6.6	1.33	0.94	6.6	1.58	1.10	6.5	1.88	1.27

D. Model with Insurance Tax-System

Notes: A negative sign indicate gains and a positive sign losses.

Tables

			A	A. Model With				
	$\omega = 1$.0	$\omega = 1$.3	$\omega = 2.0$			
γ	λ_0	λ_1	λ_0	λ_1	λ_0	λ_1		
0.74	-0.69	-0.21	-0.55	-0.07	-0.38	0.11		
0.73	-1.38	-0.43	-1.12	-0.16	-0.77	0.22		
0.72	-2.09	-0.67	-1.68	-0.25	-1.16	0.31		
0.71	-2.81	-0.91	-2.26	-0.36	-1.56	0.39		
0.70	-3.54	-1.17	-2.85	-0.48	-1.97	0.45		

Table 3. Welfare Consequences of Migration ($\rho = 0.25$) A. Model Without Taxes

B. Model With Fixed Income Tax

·····				ouce .	VIUII I IACU I		
	$\omega = 1$.0	$\omega = 1$.3	$\omega = 2.0$		
γ	λ_0	λ_1	λ_0	λ_1	λ_0	λ_1	
0.74	-0.69	-0.08	-0.55	0.10	-0.38	0.38	
0.73	-1.38	-0.17	-1.12	0.18	-0.77	0.74	
0.72	-2.09	-0.29	-1.68	0.24	-1.16	1.07	
0.71	-2.81	-0.42	-2.26	0.29	-1.56	1.37	
0.70	-3.54	-0.57	-2.85	0.31	-1.97	1.66	

C. Egalitarian Tax-System

				·	<u> </u>		· · · · · · · · · · · · · · · · · · ·		
	$\omega = 1.0$			$\omega = 1.3$			$\omega = 2.0$		
γ	τ	λ_0	λ_1	τ	λ_0	λ_1	τ	λ_0	λ_1
0.74	5.6	0.25	0.20	5.5	0.27	0.22	5.4	0.30	0.27
0.73	6.2	0.54	0.43	6.0	0.57	0.47	5.8	0.62	0.55
0.72	6.8	0.85	0.68	6.5	0.88	0.74	6.3	0.96	0.85
0.71	7.4	1.18	0.96	7.1	1.22	1.03	6.7	1.32	1.17
0.70	8.0	1.54	1.26	7.6	1.58	1.34	7.2	1.69	1.51

	$\omega =$	$\omega = 1.0$			$\omega = 1.3$			$\omega = 2.0$		
γ	τ	λ_0	λ_1	τ	λ_0	λ_1	au	λ_0	λ_1	
0.74	5.6	0.29	0.21	5.5	0.33	0.23	5.5	0.37	0.26	
0.73	6.2	0.64	0.46	6.1	0.70	0.50	5.9	0.78	0.53	
0.72	6.9	1.04	0.75	6.7	1.12	0.79	6.4	1.22	0.84	
0.71	7.6	1.50	1.08	7.3	1.59	1.12	7.0	1.70	1.17	
0.70	8.3	2.02	1.45	8.0	2.11	1.49	7.5	2.22	1.52	

D. Insurance Tax-System

Notes: A negative sign indicate gains and a positive one losses.

~

- Maria Cristina Marcuzzo [1985] "Yoan Violet Robinson (1903-1983)", pp. 134
- Sergio Lugaresi [1986] "Le imposte nelle teorie del sovrappiù", pp. 26
- Massimo D'Angelillo e Leonardo Paggi [1986] "PCI e socialdemocrazie europee. Quale riformismo?", pp. 158
- Gian Paolo Caselli e Gabriele Pastrello [1986] "Un suggerimento hobsoniano su terziario ed occupazione: il caso degli Stati Uniti 1960/1983", pp. 52
- Paolo Bosi e Paolo Silvestri [1986] "La distribuzione per aree disciplinari dei fondi destinati ai Dipartimenti, Istituti e Centri dell'Università di Modena: una proposta di riforma", pp. 25
- Marco Lippi [1986] "Aggregations and Dynamic in One-Equation Econometric Models", pp. 64
- Paolo Silvestri [1986] "Le tasse scolastiche e universitarie nella Legge Finanziaria 1986", pp. 41
- Mario Forni [1986] "Storie familiari e storie di proprietà. Itinerari sociali nell'agricoltura italiana del dopoguerra", pp. 165
- Sergio Paba [1986] "Gruppi strategici e concentrazione nell'industria europea degli elettrodomestici bianchi", pp. 56
- Nerio Naldi [1986] "L'efficienza marginale del capitale nel breve periodo", pp. 54
- 11. Fernando Vianello [1986] "Labour Theory of Value", pp. 31
- 12. Piero Ganugi [1986] "Risparnio forzato e politica monetaria negli economisti italiani tra le due guerre", pp. 40
- Maria Cristina Marcuzzo e Annalisa Rosselli [1986] "The Theory of the Gold Standard and Ricardo's Standard Comodity", pp. 30
- Giovanni Solinas [1986] "Mercati del lavoro locali e carriere di lavoro giovanili", pp. 66
- Giovanni Bonifati [1986] "Saggio dell'interesse e domanda effettiva. Osservazioni sul cap. 17 della General Theory", pp. 42
- Marina Murat [1986] "Betwin old and new classical macroeconomics: notes on Lejonhufvud's notion of full information equilibrium", pp. 20
- 17. Sebastiano Brusco e Giovanni Solinas [1986] "Mobilità occupazionale e disoccupazione in Emilia Romagna", pp. 48
- 18. Mario Forni [1986] "Aggregazione ed esogeneità", pp. 13
- Sergio Lugaresi [1987] "Redistribuzione del reddito, consumi e occupazione", pp. 17
- 20. Fiorenzo Sperotto [1987] "L'immagine neopopulista di mercato debole nel primo dibattito sovietico sulla pianificazione", pp. 34
- M. Cecilia Guerra [1987] "Benefici tributari nel regime misto per i dividendi proposto dalla commissione Sarcinelli: una nota critica", pp. 9
- Leonardo Paggi [1987] "Contemporary Europe and Modern America: Theories of Modernity in Comparative Perspective", pp. 38
- 23. Fernando Vianello [1987] "A Critique of Professor Goodwin's 'Critique of Sraffa'", pp. 12
- Fernando Vianello [1987] "Effective Demand and the Rate of Profits. Some Thoughts on Marx, Kalecki and Sraffa", pp. 41
- Anna Maria Sala [1987] "Banche e territorio. Approccio ad un tema geografico-economico", pp. 40
- Enzo Mingione e Giovanni Mottura [1987] "Fattori di trasformazione e nuovi profili sociali nell'agricoltura italiana: qualche elemento di discussione", pp. 36
- 27. Giovanna Procacci [1988] "The State and Social Control in Italy During the First World War", pp. 18

- Massimo Matteuzzi e Annamaria Simonazzi [1988] "Il debito pubblico", pp. 62
- Maria Cristina Marcuzzo (a cura di) [1988] "Richard F. Kahn. A discipline of Keynes", pp. 118
- Paolo Bosi [1988] "MICROMOD. Un modello dell'economia italiana per la didattica della politica fiscale", pp. 34
- Paolo Bosi [1988] "Indicatori della politica fiscale. Una rassegna e un confronto con l'aiuto di MICROMOD", pp. 25
- Giovanna Procacci [1988] "Protesta popolare e agitazioni operaie in Italia 1915-1918", pp. 45
- Margherita Russo [1988] "Distretto Industriale e servizi. Uno studio dei trasporti nella produzione e nella vendita delle piastrelle", pp. 157
- Margherita Russo [1988] "The effect of technical change on skill requirements: an empirical analysis", pp. 28
- 35. Carlo Grillenzoni [1988] "Identification, estimations of multivariate transfer functions", pp. 33
- 36. Nerio Naldi [1988] "'Keynes' concept of capital", pp. 40
- 37. Andrea Ginzburg [1988] "locomotiva Italia?", pp. 30
- Giovanni Mottura [1988] "La 'persistenza' secolare. Appunti su agricoltura contadina ed agricoltura familiare nelle società industriali", pp. 40
- Giovanni Mottura [1988] "L'anticamera dell'esodo. I contadini italiani della 'restaurazione contrattuale' fascista alla riforma fondiaria", pp. 40
- Leonardo Paggi [1988] "Americanismo e riformismo. La socialdemocrazia europea nell'economia mondiale aperta", pp. 120
- Annamaria Simonazzi [1988] "Fenomeni di isteresi nella spiegazione degli alti tassi di interesse reale", pp. 44
- Antonietta Bassetti [1989] "Analisi dell'andamento e della casualità della borsa valori", pp. 12
- Giovanna Procacci [1989] "State coercion and worker solidarity in Italy (1915-1918): the moral and political content of social unrest", pp. 41
- Carlo Alberto Magni [1989] "Reputazione e credibilità di una minaccia in un gioco bargaining", pp. 56
- 45. Giovanni Mottura [1989] "Agricoltura familiare e sistema agroalimentare in Italia", pp. 84
- Mario Forni [1989] "Trend, Cycle and 'Fortuitous cancellation': a Note on a Paper by Nelson and Plosser", pp. 4
- Paolo Bosi, Roberto Golinelli, Anna Stagni [1989] "Le origini del debito pubblico e il costo della stabilizzazione", pp. 26
- Roberto Golinelli [1989] "Note sulla struttura e sull'impiego dei modelli macroeconometrici", pp. 21
- 49. Marco Lippi [1989] "A Shorte Note on Cointegration and Aggregation", pp. 11
- Gian Paolo Caselli e Gabriele Pastrello [1989] "The Linkage between Tertiary and Industrial Sector in the Italian Economy: 1951-1988. From an External Dependence to an International One", pp. 40
- Gabriele Pastrello [1989] "Francois quesnay: dal Tableau Zig-zag al Tableau Formule: una ricostruzione", pp. 48
- 52. Paolo Silvestri [1989] "Il bilancio dello stato", pp. 34
- Tim Mason [1990] "Tre seminari di storia sociale contemporanea", pp. 26
- Michele Lalla [1990] "The Aggregate Escape Rate Analysed throught the Queueing Model", pp. 23
- Paolo Silvestri [1990] "Sull'autonomia finanziaria dell'università", pp. 11

- Paola Bertolini, Enrico Giovannetti [1990] "Uno studio di 'filiera' nell'agroindustria. Il caso del Parmigiano Reggiano", pp. 164
- Paolo Bosi, Roberto Golinelli, Anna Stagni [1990] "Effetti macroeconomici, settoriali e distributivi dell'armonizzazione dell'IVA", pp. 24
- Michele Lalla [1990] "Modelling Employment Spells from Emilia Labour Force Data", pp. 18
- Andrea Ginzburg [1990] "Politica Nazionale e commercio internazionale", pp. 22
- 60. Andrea Giommi [1990] "La probabilità individuale di risposta nel trattamento dei dati mancanti", pp. 13
- Gian Paolo Caselli e Gabriele Pastrello [1990] "The service sector in planned economies. Past experiences and future prospectives", pp. 32
- 62. Giovanni Solinas [1990] "Competenze, grandi industrie e distretti industriali, Il caso Magneti Marelli", pp. 23
- Andrea Ginzburg [1990] "Debito pubblico, teorie monetarie e tradizione civica nell'Inghilterra del Settecento", pp. 30
- Mario Forni [1990] "Incertezza, informazione e mercati assicurativi: una rassegna", pp. 37
- 65. Mario Forni [1990] "Misspecification in Dynamic Models", pp. 19
- Gian Paolo Caselli e Gabriele Pastrello [1990] "Service Sector Growth in CPE's: An Unsolved Dilemma", pp. 28
- 67. Paola Bertolini [1990] "La situazione agro-alimentare nei paesi ad economia avanzata", pp. 20
- Paola Bertolini [1990] "Sistema agro-alimentare in Emilia Romagna ed occupazione", pp. 65
- Enrico Giovannetti [1990] "Efficienza ed innovazione: il modello "fondi e flussi" applicato ad una filiera agro-industriale", pp. 38
- 70. Margherita Russo [1990] "Cambiamento tecnico e distretto industriale: una verifica empirica", pp. 115
- Margherita Russo [1990] "Distretti industriali in teoria e in pratica: una raccolta di saggi", pp. 119
- 72. Paolo Silvestri [1990] " La Legge Finanziaria. Voce dell'enciclopedia Europea Garzanti", pp. 8
- 73. Rita Paltrinieri [1990] "La popolazione italiana: problemi di oggi e di domani", pp. 57
- 74. Enrico Giovannetti [1990] "Illusioni ottiche negli andamenti delle Grandezze distributive: la scala mobile e l'appiattimento' delle retribuzioni in una ricerca", pp. 120
- Enrico Giovannetti [1990] "Crisi e mercato del lavoro in un distretto industriale: il bacino delle ceramiche. Sez I", pp. 150
- Enrico Giovannetti [1990] "Crisi e mercato del lavoro in un distretto industriale: il bacino delle ceramiche. Sez. II", pp. 145
- Antonietta Bassetti e Costanza Torricelli [1990] "Una riqualificazione dell'approccio bargaining alla selezioni di portafoglio", pp. 4
- Antonietta Bassetti e Costanza Torricelli [1990] "Il portafoglio ottimo come soluzione di un gioco bargaining", pp. 15
- 79. Mario Forni [1990] "Una nota sull'errore di aggregazione", pp. 6
- Francesca Bergamini [1991] "Alcune considerazioni sulle soluzioni di un gioco bargaining", pp. 21
- Michele Grillo e Michele Polo [1991] "Political Exchange and the allocation of surplus: a Model of Two-party competition", pp. 34
- Gian Paolo Caselli e Gabriele Pastrello [1991] "The 1990 Polish Recession: a Case of Truncated Multiplier Process", pp. 26
- Gian Paolo Caselli e Gabriele Pastrello [1991] "Polish firms: Pricate Vices Pubblis Virtues", pp. 20
- Sebastiano Brusco e Sergio Paba [1991] "Connessioni, competenze e capacità concorrenziale nell'industria della Sardegna", pp. 25

- Claudio Grimaldi, Rony Hamaui, Nicola Rossi [1991] "Non Marketable assets and hauseholds' Portfolio Choice: a Case of Study of Italy", pp. 38
- Giulio Righi, Massimo Baldini, Alessandra Brambilla [1991] "Le misure degli effetti redistributivi delle imposte indirette: confronto tra modelli alternativi", pp. 47
- Roberto Fanfani, Luca Lanini [1991] "Innovazione e servizi nello sviluppo della meccanizzazione agricola in Italia", pp. 35
- Antonella Caiumi e Roberto Golinelli [1992] "Stima e applicazioni di un sistema di domanda Almost Ideal per l'economia italiana", pp. 34
- Maria Cristina Marcuzzo [1992] "La relazione salari-occupazione tra rigidità reali e rigidità nominali", pp. 30
- Mario Biagioli [1992] "Employee financial participation in enterprise results in Italy", pp. 50
- Mario Biagioli [1992] "Wage structure, relative prices and international competitiveness", pp. 50
- Paolo Silvestri e Giovanni Solinas [1993] "Abbandoni, esiti e carriera scolastica. Uno studio sugli studenti iscritti alla Facoltà di Economia e Commercio dell'Università di Modena nell'anno accademico 1990/1991", pp. 30
- Gian Paolo Caselli e Luca Martinelli [1993] "Italian GPN growth 1890-1992: a unit root or segmented trend representatin?", pp. 30
- 94. Angela Politi [1993] "La rivoluzione fraintesa. I partigiani emiliani tra liberazione e guerra fredda, 1945-1955", pp. 55
- Alberto Rinaldi [1993] "Lo sviluppo dell'industria metalmeccanica in provincia di Modena: 1945-1990", pp. 70
- Paolo Emilio Mistrulli [1993] "Debito pubblico, intermediari finanziari e tassi d'interesse: il caso italiano", pp. 30
- Barbara Pistoresi [1993] "Modelling disaggregate and aggregate labour demand equations. Cointegration analysis of a labour demand function for the Main Sectors of the Italian Economy: 1950-1990", pp. 45
- Giovanni Bonifati [1993] "Progresso tecnico e accumulazione di conoscenza nella teoria neoclassica della crescita endogena. Una analisi critica del modello di Romer", pp. 50
- Marcello D'Amato e Barbara Pistoresi [1994] "The relationship(s) among Wages, Prices, Unemployment and Productivity in Italy", pp. 30
- Mario Forni [1994] "Consumption Volatility and Income Persistence in the Permanent Income Model", pp. 30
- Barbara Pistoresi [1994] "Using a VECM to characterise the relative importance of permanent and transitority components", pp. 28
- 102. Gian Paolo Caselli and Gabriele Pastrello [1994] "Polish recovery form the slump to an old dilemma", pp. 20
- Sergio Paba [1994] "Imprese visibili, accesso al mercato e organizzazione della produzione", pp. 20
- Giovanni Bonifati [1994] "Progresso tecnico, investimenti e capacità produttiva", pp. 30
- Giuseppe Marotta [1994] "Credit view and trade credit: evidence from Italy", pp. 20
- Margherita Russo [1994] "Unit of investigation for local economic development policies", pp. 25
- Luigi Brighi [1995] "Monotonicity and the demand theory of the weak axioms", pp. 20
- Mario Forni e Lucrezia Reichlin [1995] "Modelling the impact of technological change across sectors and over time in manufactoring", pp. 25
- Marcello D'Amato and Barbara Pistoresi [1995] "Modelling wage growth dynamics in Italy: 1960-1990", pp. 38
- Massimo Baldini [1995] "INDIMOD. Un modello di microsimulazione per lo studio delle imposte indirette", pp. 37

- Paolo Bosi [1995] "Regionalismo fiscale e autonomia tributaria: l'emersione di un modello di consenso", pp. 38
- Massimo Baldini [1995] "Aggregation Factors and Aggregation Bias in Consumer Demand", pp. 33
- Costanza Torricelli [1995] "The information in the term structure of interest rates. Can stocastic models help in resolving the puzzle?" pp. 25
- 114. Margherita Russo [1995] "Industrial complex, pôle de développement, distretto industriale. Alcune questioni sulle unità di indagine nell'analisi dello sviluppo." pp. 45
- Angelika Moryson [1995] "50 Jahre Deutschland. 1945 1995" pp. 21
- 116. Paolo Bosi [1995] "Un punto di vista macroeconomico sulle caratteristiche di lungo periodo del nuovo sistema pensionistico italiano." pp. 32
- 117. Gian Paolo Caselli e Salvatore Curatolo [1995] "Esistono relazioni stimabili fra dimensione ed efficienza delle istituzioni e crescita produttiva? Un esercizio nello spirito di D.C. North." pp. 11
- 118. Mario Forni e Marco Lippi [1995] "Permanent income, heterogeneity and the error correction mechanism." pp. 21
- 119. Barbara Pistoresi [1995] "Co-movements and convergence in international output. A Dynamic Principal Components Analysis" pp. 14
- Mario Forni e Lucrezia Reichlin [1995] "Dynamic common factors in large cross-section" pp. 17
- Giuseppe Marotta [1995] "Il credito commerciale in Italia: una nota su alcuni aspetti strutturali e sulle implicazioni di politica monetaria" pp. 20
- 122. Giovanni Bonifati [1995] "Progresso tecnico, concorrenza e decisioni di investimento: una analisi delle determinanti di lungo periodo degli investimenti" pp. 25
- Giovanni Bonifati [1995] "Cambiamento tecnico e crescita endogena: una valutazione critica delle ipotesi del modello di Romer" pp. 21
- 124. Barbara Pistoresi e Marcello D'Amato [1995] "La riservatezza del banchiere centrale è un bene o un male? "Effetti dell'informazione incompleta sul benessere in un modello di politica monetaria." pp. 32
- Barbara Pistoresi [1995] "Radici unitarie e persistenza: l'analisi univariata delle fluttuazioni economiche." pp. 33
- Barbara Pistoresi e Marcello D'Amato [1995] "Co-movements in European real outputs" pp. 20
- Antonio Ribba [1996] "Ciclo economico, modello lineare-stocastico, forma dello spettro delle variabili macroeconomiche" pp. 31
- 128. Carlo Alberto Magni [1996] "Repeatable and una tantum real options a dynamic programming approach" pp. 23
- 129. Carlo Alberto Magni [1996] "Opzioni reali d'investimento e interazione competitiva: programmazione dinamica stocastica in optimal stopping" pp. 26
- Carlo Alberto Magni [1996] "Vaghezza e logica fuzzy nella valutazione di un'opzione reale" pp. 20
- Giuseppe Marotta [1996] "Does trade credit redistribution thwart monetary policy? Evidence from Italy" pp. 20
- Mauro Dell'Amico e Marco Trubian [1996] "Almost-optimal solution of large weighted equicut problems" pp. 30
- Carlo Alberto Magni [1996] "Un esempio di investimento industriale con interazione competitiva e avversione al rischio" pp. 20
- 134. Margherita Russo, Peter Börkey, Emilio Cubel, François Lévêque, Francisco Mas [1996] "Local sustainability and competitiveness: the case of the ceramic tile industry" pp. 66
- Margherita Russo [1996] "Camionetto tecnico e relazioni tra imprese" pp. 190
- David Avra Lane, Irene Poli, Michele Lalla, Alberto Roverato [1996] "Lezioni di probabilità e inferenza statistica" pp. 288

- David Avra Lane, Irene Poli, Michele Lalla, Alberto Roverato [1996] "Lezioni di probabilità e inferenza statistica - Esercizi svolti -" pp. 302
- Barbara Pistoresi [1996] "Is an Aggregate Error Correction Model Representative of Disaggregate Behaviours? An example" pp. 24
- 139. Luisa Malaguti e Costanza Torricelli [1996] "Monetary policy and the term structure of interest rates", pp. 30
- Mauro Dell'Amico, Martine Labbé, Francesco Maffioli [1996] "Exact solution of the SONET Ring Loading Problem", pp. 20
- 141. Mauro Dell'Amico, R.J.M. Vaessens [1996] "Flow and open shop scheduling on two machines with transportation times and machineindependent processing times in NP-hard, pp. 10
- M. Dell'Amico, F. Maffioli, A. Sciomechen [1996] "A Lagrangean Heuristic for the Pirze Collecting Travelling Salesman Problem", pp. 14
- Massimo Baldini [1996] "Inequality Decomposition by Income Source in Italy - 1987 - 1993", pp. 20
- Graziella Bertocchi [1996] "Trade, Wages, and the Persistence of Underdevelopment" pp. 20
- 145. Graziella Bertocchi and Fabio Canova [1996] "Did Colonization matter for Growth? An Empirical Exploration into the Historical Causes of Africa's Underdevelopment" pp. 32
- 146. Paola Bertolini [1996] "La modernization de l'agricolture italienne et le cas de l'Emilie Romagne" pp. 20
- Enrico Giovannetti [1996] "Organisation industrielle et développement local: le cas de l'agroindutrie in Emilie Romagne" pp. 18
- 148. Maria Elena Bontempi e Roberto Golinelli [1996] "Le determinanti del leverage delle imprese: una applicazione empirica ai settori industriali dell'economia italiana" pp. 31
- 149. Paola Bertolini [1996] "L'agriculture et la politique agricole italienne face aux recents scenarios", pp. 20
- 150. Enrico Giovannetti [1996] "Il grado di utilizzo della capacità produttiva come misura dei costi di transizione. Una rilettura di 'Nature of the Firm' di R. Coase", pp. 65
- Enrico Giovannetti [1996] "Il Iº ciclo del Diploma Universitario Economia e Amministrazione delle Imprese", pp. 25
- 152. Paola Bertolini, Enrico Giovannetti, Giulia Santacaterina [1996] "Il Settore del Verde Pubblico. Analisi della domanda e valutazione economica dei benefici", pp. 35
- Giovanni Solinas [1996] "Sistemi produttivi del Centro-Nord e del Mezzogiorno. L'industria delle calzature", pp. 55
- Tindara Addabbo [1996] "Married Women's Labour Supply in Italy in a Regional Perspective", pp. 85
- 155. Paolo Silvestri, Giuseppe Catalano, Cristina Bevilacqua [1996] "Le tasse universitarie e gli interventi per il diritto allo studio: la prima fase di applicazione di una nuova normativa" pp. 159
- Sebastiano Brusco, Paolo Bertossi, Margherita Russo [1996] "L'industria dei rifiuti urbani in Italia", pp. 25
- 157. Paolo Silvestri, Giuseppe Catalano [1996] "Le risorse del sistema universitario italiano: finanziamento e governo" pp. 400
- Carlo Alberto Magni [1996] "Un semplice modello di opzione di differimento e di vendita in ambito discreto", pp. 10
- Tito Pietra, Paolo Siconolfi [1996] "Fully Revealing Equilibria in Sequential Economies with Asset Markets" pp. 17
- Tito Pietra, Paolo Siconolfi [1996] "Extrinsic Uncertainty and the Informational Role of Prices" pp. 42
- 161. Paolo Bertella Farnetti [1996] "Il negro e il rosso. Un precedente non esplorato dell'integrazione afroamericana negli Stati Uniti" pp. 26
- 162. David Lane [1996] "Is what is good for each best for all? Learning from others in the information contagion model" pp. 18

- Antonio Ribba [1996] "A note on the equivalence of long-run and short-run identifying restrictions in cointegrated systems" pp. 10
- 164. Antonio Ribba [1996] "Scomposizioni permanenti-transitorie in sistemi cointegrati con una applicazione a dati italiani" pp. 23
- Mario Forni, Sergio Paba [1996] "Economic Growth, Social Cohesion and Crime" pp. 20
- Mario Forni, Lucrezia Reichlin [1996] "Let's get real: a factor analytical approch to disaggregated business cycle dynamics" pp. 25
- 167. Marcello D'Amato e Barbara Pistoresi [1996] "So many Italies: Statistical Evidence on Regional Cohesion" pp. 31
- Elena Bonfiglioli, Paolo Bosi, Stefano Toso [1996] "L'equità del contributo straordinario per l'Europa" pp. 20
- 169. Graziella Bertocchi, Michael Spagat [1996] "Il ruolo dei licei e delle scuole tecnico-professionali tra progresso tecnologico, conflitto sociale e sviluppo economico" pp. 37
- 170. Gianna Boero, Costanza Torricelli [1997] "The Expectations Hypothesis of the Term Structure of Interest Rates: Evidence for Germany" pp. 15
- Mario Forni, Lucrezia Reichlin [1997] "National Policies and Local Economies: Europe and the US" pp. 22
- 172. Carlo Alberto Magni [1997] "La trappola del Roe e la tridimensionalità del Van in un approccio sistemico", pp. 16
- 173. Mauro Dell'Amico [1997] "A Linear Time Algorithm for Scheduling Outforests with Communication Delays on Two or Three Processor"pp 18
- 174. Paolo Bosi [1997] "Aumentare l'età pensionabile fa diminuire la spesa pensionistica? Ancora sulle caratteristiche di lungo periodo della riforma Dini" pp 13
- 175. Paolo Bosi e Massimo Matteuzzi [1997] "Nuovi strumenti per l'assistenza sociale" pp 31
- Mauro Dell'Amico, Francesco Maffioli e Marco Trubian [1997] "New bounds for optium traffic assignment in satellite communication" pp 21
- 177. Carlo Alberto Magni [1997] "Paradossi, inverosimiglianze e contraddizioni del Van: operazioni certe" pp 9
- 178. Barbara Pistoresi e Marcello D'Amato [1997] "Persistence of relative unemployment rates across italian regions" pp 25
- Margherita Russo, Franco Cavedoni e Riccardo Pianesani [1997] "Le spese ambientali dei Comuni in provincia di Modena, 1993-1995" pp. 23
- Gabriele Pastrello [1997] "Time and Equilibrium, Two Elisive Guests in the Keynes-Hawtrey-Robertson Debate in the Thirties" pp. 25
- 181. Luisa Malaguti e Costanza Torricelli [1997] "The Interaction Between Monetary Policy and the Expectation Hypothesis of the Term Structure of Interest rates in a N-Period Rational Expectation Model" pp. 27
- Mauro Dell'Amico [1997] "On the Continuous Relaxation of Packing Problems - Technical Note" pp. 8
- 183. Stefano Bordoni [1997] "Prova di Idoneità di Informatica Dispensa Esercizi Excel 5" pp 49
- Francesca Bergamini e Stefano Bordoni [1997] "Una verifica empirica di un nuovo metodo di selezione ottima di portafoglio" pp 22
- 185. Gian Paolo Caselli e Maurizio Battini [1997] "Following the tracks of atkinson and micklewright the changing distribution of income and earnings in poland from 1989 to 1995", pp 21
- Mauro Dell'Amico e Francesco Maffioli [1997] "Combining Linear and Non-Linear Objectives in Spanning Tree Problems" pp 21
- Gianni Ricci e Vanessa Debbia [1997] "Una soluzione evolutiva in un gioco differenziale di lotta di classe" pp14
- Fabio Canova e Eva Ortega [1997] "Testing Calibrated General Equilibrium Model" pp 34

- 189. Fabio Canova [1997] "Does Detrending Matter for the Determination of the Reference Cycle and the Selection of Turning Points?" pp35
- 190. Fabio Canova e Gianni De Nicolò [1997] "The Equity Premium and the Risk Free Rate: A Cross Country, Cross Maturity Examination" pp 41
- Fabio Canova e Angel J. Ubide [1997] "International Business Cycles, Financial Market and Household Production" pp 32
- Fabio Canova e Gianni De Nicolò [1997] "Stock Returns, Term Structure, Inflation and Real Activity: An International Perspective" pp 33
- Fabio Canova e Morten Ravn [1997] "The Macroeconomic Effects of German Unification: Real Adjustmentsand the Welfare State" pp 34
- 194. Fabio Canova [1997] "Detrending and Business Cycle Facts" pp 40